

# The Empirics of Economic Growth Over Time and Across Nations: A Unified Growth Perspective

Matteo Cervellati (University of Bologna and CEPR) Gerrit Meyerheim (LMU Munich) Uwe Sunde (LMU Munich, CEPR, CESifo and IZA)

Discussion Paper No. 339

September 29, 2022

# The Empirics of Economic Growth Over Time and Across Nations: A Unified Growth Perspective \*

Matteo Cervellati University of Bologna CEPR

Gerrit Meyerheim

University of Munich

University Munich CEPR, CESifo, IZA

Uwe Sunde

#### Abstract

This research develops an expanded unified growth theory that incorporates the endogenous accumulation of physical capital, population, human capital, and technology. The model incorporates a complementarity between physical capital and human capital and can be extended to a multi-country setting with international technology diffusion. The analytical characterization of the mechanisms behind the observed patterns of long-run growth and comparative development delivers a consistent explanation for a large set of seemingly unrelated empirical facts. A quantitative multi-country version of the model matches various empirical regularities of long-run growth dynamics and comparative development patterns that have previously been studied in isolation. The findings also shed new light on the role of the demographic transition for convergence patterns, the specification of cross-country growth regressions, technology spillovers, and the secular stagnation debate.

JEL-classification: O47, O11, O15, E24

*Keywords:* 

Unified Growth, Long-run Development, Demographic Transition, Secular Stagnation

<sup>\*</sup>The authors wish to thank Oded Galor, Stephen Broadberry, Lukas Rosenberger, Thomas Steger, Jaume Ventura, and seminar participants at Humboldt University Berlin, the University of Leipzig, LMU Munich, and the CEPR Conference "Accounting For The Wealth Of Nations: History And Theory" at the University of Southern Denmark for helpful comments and suggestions. Daniel Endler and Johannes Lohwasser provided excellent research assistance.

Uwe Sunde gratefully acknowledges funding by the by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) – Project 491578970; and through CRC TRR 190 (Project 280092119, A07). Contacts: Matteo Cervellati, University of Bologna, Piazza Scaravilli 2, 40126 Bologna, Italy, Email: m.cervellati@unibo.it; Gerrit Meyerheim, University of Munich (LMU), Geschwister-Scholl-Platz 1, 80539 Munich, Germany, Email: gerrit.meverheim@econ.lmu.de; Uwe Sunde (corresponding author), University of Munich (LMU), Geschwister-Scholl-Platz 1, 80539 Munich, Germany, Email: uwe.sunde@lmu.de, Tel. +49 89 2180 1280.

## 1 Introduction

Over the past decades, the literature on economic growth and economic history has documented a considerable body of empirical patterns of long-run growth that relate to the dynamics of income, demographic variables, production factors, and factor returns. The most evident of these regularities is the secular acceleration of economic well-being over human history. While traditional models of balanced economic growth had limited success in accounting for this acceleration, the development of unified growth theory has shifted the focus to the mechanics behind the transition from Malthusian stagnation to sustained economic growth and the deep roots of comparative development (see Galor, 2011, 2022). While the last two decades have seen remarkable progress in the understanding of the theoretical mechanisms behind this transition, the applicability of quantitative unified growth models to empirical patterns of long-run growth dynamics and comparative development constitutes a relatively underexplored area. This is related, on the one hand, to the discrepancy in the emphasis given to physical capital accumulation in empirical studies of growth, which is not shared by most unified growth models with their focus on education and fertility. On the other hand, studying comparative development requires an explicit consideration of cross-country heterogeneity and interdependencies, in particular, of the diffusion of technology across countries.

This paper develops an expanded unified growth framework that makes progress in several dimensions. We present an analytically tractable unified growth model that extends the focus beyond endogenous fertility and education by accounting explicitly for savings and physical capital accumulation. In particular, this allows incorporating complementarities between physical capital and human capital, an aspect that has received relatively little attention in existing work on long-run growth. We then show that a quantitative version of the model can go a long way in matching the empirical patterns of long-run growth, including the behavior of the state variables physical capital, human capital, population, and technology based on an up-to-date account of the facts of longrun growth observed in historical time series data. An extension of the model to the multi-country context accounts for interdependencies in the diffusion of technology. The comparison of the simulated data of this extension to actual cross-country panel data shows that the quantitative unified growth model can match a variety of non-targeted data moments. In addition, the findings provide new insights for standard cross-country growth empirics and the patterns of technology diffusion.

Section 2 presents a tractable unified growth model that combines the standard components of neo-classical growth models with the mechanisms and insights highlighted by unified growth theory. In particular, we consider a two-sector framework that features complementarities between human capital and physical capital and that delivers endogenous dynamics of population, physical capital, and human capital. The model delivers analytical predictions about the demographic transition as the critical turning point in the dynamics of all state variables and about non-linear dynamics in all core dimensions that are consistent with the empirical facts. In particular, technological change induces the accumulation of physical capital, with the consequence of moderate growth in income per capita even prior to the demographic transition, when the economy is still characterized by high fertility, low education, and Malthusian dynamics. The accumulation of physical capital in combination with a capital-skill complementarity eventually induces an expansion of human capital that is associated with a reduction in fertility. These transition dynamics have implications for the dynamics of skill premia, factor income shares, and factor returns.

Section 3 investigates the consistency of a quantitative version of the model with the empirical facts of long-run growth, closing a gap between theoretical models and quantitative empirical facts in the literature. In this section, we focus on the development trajectory of a single economy and calibrate the model to data moments for England, paralleling the usual approach in unified growth and historical studies. The results document the ability of the model to replicate the facts of long-run growth, including the secular acceleration of income growth, which coincides with the demographic transition that marks the beginning of the education expansion. The model also replicates several empirical patterns that have been documented only recently, including the temporary overshooting of growth rates above the rate of sustained modern growth, the secular stagnation during the later phase of the transition, the corresponding non-monotonic dynamics of skill premia and factor income shares, and the relatively stable dynamics of the returns to physical capital. In particular, the non-monotonic dynamics of capital intensity per worker and of the skill premium, both of which tend to decline during the early phases of development but increase during the later stages of development, can be rationalized by the interactions between human capital and physical capital, without requiring specific assumptions about skill-biased technological change or inter-temporal spillovers. Moreover, the joint rapid expansion of human capital and physical capital after the onset of the demographic transition leads to a temporary overshooting of income growth above the growth rate along the balanced growth path. The analysis also shows that the simulated model closely matches actual time series in various dimensions that have not been targeted.

Section 4 extends the model to a multi-country setting and explicitly incorporates the process of international technology diffusion. The only source of heterogeneity across countries is variation in a single initial condition that governs the timing of the demographic transition, otherwise the calibrated model is identical to the baseline calibration for England. The endogenous improvement of the technological frontier displays a sharp acceleration when the forerunner country undergoes the economic and demographic transition. Follower countries benefit from this acceleration, but only after the onset of their own demographic transition that corresponds to the expansion in human capital and thereby enables the adoption of technologies from the frontier. As a result, productivity initially displays a rapid divergence across countries as the frontier accelerates while the spillovers to follower countries are still limited. Eventually, follower countries achieve even faster income growth than forerunners once they undergo their demographic transition. The analysis documents that the simulated data deliver comparative development patterns that are consistent with those observed in the data, including the variation in income and productivity across countries, cross-country differences in capital intensity and capital returns, and in the variability of growth rates relative to the technology frontier.

Section 5 makes further progress by exploring the similarity between simulated and actual data patterns through the lens of standard growth empirics. In particular, we document that a synthetic panel data set produced by a simulation of a multi-country version of the quantitative model delivers reduced-form estimation results that resemble those obtained from reduced-form empirical growth regressions. Extending the model to human capital-dependent technology adoption from the endogenously evolving world frontier allows us to explore the implications of the mechanism for the facts on technology diffusion. Regression results deliver new evidence on the drivers of cross-country growth and show that the demographic transition marks a critical turning point for technology adoption both at the extensive and at the intensive margin.

Taken together, the analytical, quantitative, and empirical results provide a proof of concept for both the quantitative and the empirical validity of the generalized unified paradigm that is based on the close interplay between the demographic and economic transition. The expanded unified growth framework sheds new light on the mechanisms that give rise to the patterns of comparative development, provides a link to the results of empirical growth regressions, and documents that the proposed model can go a long way in providing a consistent, unified explanation for a large variety of seemingly unrelated facts of long-run growth.

**Contribution to the Literature.** This research contributes to several strands of the literature. In terms of theory, our model incorporates prototypical elements and trade-offs from the literature on neo-classical growth and unified growth within a single, coherent framework. The model incorporates endogenous fertility choice and a demographic transition (as in, e.g., Galor and Weil, 2000, Strulik and Weisdorf, 2008) in a two-sector setting where sectors differ in their use of unskilled and skilled labor (as in, e.g., Doepke, 2004 or Cervellati and Sunde, 2005). While most of the unified growth literature uses a quantity-quality trade-off to illustrate that fertility reductions are ultimately the result of an increasing demand for human capital, we consider a trade-off between individuals' own education and their fertility (as in, e.g., Cervellati and Sunde, 2015). Both approaches deliver similar predictions regarding the nexus of education and fertility, but our approach

provides a more direct and parsimonious representation that facilitates the incorporation of savings. Malthusian dynamics emerge by relating production in the unskilled sector to a fixed factor (as in, e.g., Ashraf and Galor, 2011). Production in the skilled sector involves physical capital (as in, e.g., Hansen and Prescott, 2002, Bar and Leukhina, 2010, or Strulik et al., 2013). Physical capital complements skilled human capital in aggregate production (as in, e.g., Galor and Moav, 2006, or Galor, Moav and Vollrath, 2009), while resource constraints imply a trade-off between investment in education, fertility, and savings at the household level. The model complements the existing theoretical literature by presenting a mechanism that links the demographic transition and physical capital dynamics. In particular, by considering a capital-skill complementarity, the endogenous physical capital dynamics generate an increasing demand for skilled workers, which allows reconciling the predictions of unified growth theory with a variety of facts of long-run growth without relying on specific forms of inter-generational externalities.

The consideration of a unified growth framework with endogenous savings and education investments illustrates that the end of the Malthusian period not only constitutes the onset of a demographic transition with an expansion of human capital and a reduction in fertility, but also marks a crucial turning point in the accumulation of physical capital. As a result, skill premia do not exhibit a persistent decline in spite of the expansion of human capital. This provides a foundation for non-decreasing returns to human capital that have been assumed in reduced form by, e.g., Becker et al. (1990), and parallels arguments by Kaldor (1957) or Zeira (2009). Likewise, capital returns do not exhibit a pronounced decline in spite of the expansion of physical capital. The results also contribute insights to the convergence debate. Conventionally, unified growth theories predict income growth to accelerate in the context of the demographic transition and ultimately converge to the balanced growth path "from below" (see, e.g., Galor and Weil, 2000, Cervellati and Sunde, 2005). In contrast, neo-classical models of economic growth and augmented neoclassical models that account for the historical development patterns predict convergence patterns with growth rates converging to the balanced growth path "from above" (see, e.g., Mankiw, et al., 1992, Dalgaard and Strulik, 2013). Rather than convergence "from below" or "from above", the transition dynamics of our model presented below imply an acceleration of growth during the transition from slow, quasi-stagnant growth dynamics to sustained growth, with growth rates during this transition temporarily exceeding the long-run growth rate as consequence of factor accumulation. This sheds new light on the mechanisms behind empirically observed patterns of convergence (see Johnson and Papageorgiou, 2020, for a recent survey).

Our analysis also contributes to the literature of quantitative models of long-run economic and demographic dynamics that target historical data. Most of this literature focuses on fertility dynamics along the transition to modern growth and either abstracts from human capital or physical capital accumulation (see, e.g., Greenwood and Sheshadri, 2002, Kalemli-Ozcan, 2003, Doepke, 2004, Bar and Leukhina, 2010) or focuses on the longrun structural change at the level of families (Greenwood et al., 2021). Methodologically, our approach departs from this literature by providing a dynamic simulation that accounts for the different phases of development dynamics in all state variables. Technically, we proceed by calibrating the structural parameters and simulating the endogenous dynamics for the entire development path of a country given initial conditions. All state variables and factor prices evolve endogenously over time and their trajectories are compared to the evolution of the time series of the different variables, most of which are not targeted in the calibration.

The quantitative analysis goes beyond the time series perspective by considering the implications for comparative development patterns, following a similar approach as in Cervellati and Sunde (2015). This also complements recent work by Delventhal et al. (2021), who investigate the timing of the demographic transition in different countries and the role of access to technology while abstracting from endogenous take-offs and Malthusian dynamics. Compared to these works, we apply a very parsimonious methodology that focuses exclusively on the timing of the demographic transition by changing one single initial condition in otherwise identically parameterized economies. In the extended version of the model, countries are linked through an endogenous process of technology diffusion. The resulting cross-country patterns effectively represent non-targeted moments that can be used to assess the external validity of the underlying mechanism. After simulating the multi-country version of the model, we extend the scope of the quantitative analyses in the existing literature by comparing simulated and actual cross-country panel data through the lens of empirical growth regressions as a novel way to study the empirical relevance of the proposed generalized unified growth framework.

Our analysis thereby establishes a link between unified growth theory and the empirical growth literature (e.g., Mankiw, Romer, and Weil, 1992). In particular, we estimate reduced-form panel regressions using synthetic panel data generated from the multi-country version of our unified growth model. The estimation results closely resemble those obtained with cross-country panel data, without relying (explicitly or implicitly) on the notion of balanced growth or convergence. The findings complement earlier empirical work on the role of the fertility transition for cross-country income growth (see also Galor, 2012), the role of the demographic transition as the starting point of a neo-classical convergence process (Dalgaard and Strulik, 2013), and evidence for different regimes of growth before and after the demographic transition (Cervellati et al., 2019). The findings also provide a rationale for the weak evidence on the role of human capital for development and growth differences, by showing that human capital becomes important mainly after the onset of the transition and by illustrating that human capital affects development through different channels (see, e.g., Sunde and Vischer, 2015). The results complement previous evidence for the existence of important non-linearities in the growth process depending on the stage of the demographic transition (see, e.g., Cervellati and Sunde, 2011a, 2011b) by accounting for the distinct dynamics of income and population during different phases of demographic development.

By documenting a temporary overshooting of growth due to the expansion of human capital and physical capital during the early phases of the take-off, the analysis also sheds new light on the "new secular stagnation" debate and the factors that are responsible for the slow-down in growth observed in developed economies during the past decades. While this debate focused on arguments related to a delayed recovery from cyclical fluctuations (see, e.g., Teulings and Baldwin, 2014, and OECD, 2015), some scholars have emphasized the possibility of a decline in growth potential due to unfavorable dynamics in demographic composition, education, and globalisation (see, e.g., Gordon, 2012, 2014, 2016). Our results consider the dynamics of the entire transition phase and suggest that part of the observed growth stagnation in developed economies can be rationalized by the predicted growth slow-down during the later phases of the transition from stagnation to growth.

Finally, the analysis contributes to the recent empirical literature on the patterns of comparative development and technology diffusion. With only few notable exceptions such as Galor and Mountford (2008), most of the existing unified growth literature has focused on a single-country setting. Our modeling complements recent work that focused on technology spillovers from a world frontier that grows at exogenous and constant rates, see, e.g., Benhabib and Spiegel (2005) or Stokey (2015, 2021). Our model also relates to work by Lindner and Strulik (2020) that considers an endogenous world technology frontier that emerges from a global network of R&D activities and provides a rationale for growth rates of follower economies that temporarily exceed those on the balanced growth path. In contrast, our focus is on the unexplored quantitative and empirical implications of the demographic transition and the associated human capital expansion for the diffusion of technology and abstracts from an explicit modeling of R&D-based growth. Combining a simple framework of technology adoption along the lines of classic work by Nelson and Phelps (1966) with a unified growth mechanism allows us to illustrate the role of the demographic transition for the endogenous acceleration in technology dynamics in the frontier economy and for the timing and intensity of the adoption of new technologies in less developed economies. To explore the empirical relevance of this mechanism, we extend the empirical investigation of the patterns of adoption lags and intensity of use by Comin and Mestieri (2018) and explicitly account for different phases of demographic development. The results provide novel evidence that the demographic transition indeed plays a significant role for explaining the empirical patterns of technology diffusion in terms of adoption lags and intensity of use that has not been documented previously.

The remainder of the paper is structured as follows. Section 2 presents an expanded unified growth model that incorporates endogenous savings and a capital-skill complementarity and that can be extended to account for human capital externalities and to a multi-country setting. Section 3 analyzes the quantitative model dynamics and their consistency with the dynamics of long-run growth. Section 4 explores the consistency of the expanded unified growth framework with patterns of comparative development. Section 5 investigates the implications for cross-country growth empirics and technology diffusion. Section 6 concludes.

## 2 A Generalized Unified Growth Model

In this section, we develop a tractable unified growth model that accounts for the dynamic interactions between fertility, education, physical capital accumulation, and technology. While kept deliberately simple and stylized, the model incorporates the state variables that are central to the long-run dynamics of economic and demographic variables and can be used for quantitative analysis. In particular, the model combines an OLG framework that features optimal decisions about human capital, savings, and fertility with a two-sector production economy. The dynamic general equilibrium path exhibits Malthusian dynamics and an endogenous growth take-off. This take-off is associated with a demographic transition that entails an acceleration of investments in human capital and physical capital which is paralleled by a decline in fertility. After this take-off, the economy enters a regime of modern balanced growth.

### 2.1 Production

Total output of the economy in period t is produced in two sectors. The sectors employ unskilled and skilled labor inputs; sectors are denoted by U and S, respectively

$$Y_t = Y_t^U + Y_t^S \tag{1}$$

Production in the unskilled sector combines unskilled labor  $L_t$  and a fixed amount of land X, using a technology

$$Y_t^U = L_t^\alpha \left( A_t^U X \right)^{1-\alpha} \quad \text{with } \alpha \in (0,1)$$
(2)

where  $A_t^U$  denotes the level of (land-augmenting) productivity. Production in the skilled sector uses skilled labor, denoted by human capital  $H_t$ , and physical capital  $K_t$ , using a neo-classical technology

$$Y_t^S = H_t^\beta \left( A_t^S K_t \right)^{1-\beta} \quad \text{with } \beta \in (0,1)$$
(3)

with (capital-augmenting) productivity denoted by  $A_t^S$ .

To streamline the exposition, and for transparency of the mechanism and the main results, we make some simplifying assumptions. First, the homogeneous output good can be used for consumption and investment. Moreover, we assume that physical capital fully depreciates after one generation, which simplifies the law of motion of aggregate capital without altering the main insights.<sup>1</sup> Second, we assume that property rights are defined and enforced over capital and labor, while no property rights are defined over land (as in Galor and Weil, 2000). Third, we assume that factor returns are determined on competitive markets. This implies that returns to human capital and physical capital in the skilled sector are given by the respective marginal products, whereas returns to unskilled labor are set equal to the average product, such that all income produced in the unskilled sector is distributed to the unskilled workers. This allows us to abstract from having to make assumptions about land markets and bequests of land in an economy with changing cohort sizes. Factor rents are thus given by

$$w_t^L = \frac{Y_t^U}{L_t} = \left(\frac{A_t^U X}{L_t}\right)^{1-\alpha} \tag{4}$$

$$w_t^H = \frac{\partial Y_t^S}{\partial H_t} = \beta \left(\frac{A_t^S K_t}{H_t}\right)^{1-\beta} \tag{5}$$

$$1 + r_t = R_t = \frac{\partial Y_t^S}{\partial K_t} = (1 - \beta) \left(\frac{H_t}{K_t}\right)^\beta \left(A_t^S\right)^{1 - \beta} \tag{6}$$

## 2.2 Optimal Education, Fertility, and Savings

Households are modeled as in a standard overlapping generations framework. In period t, a new generation of size  $L_t$  enters the economy. This generation consists of identical individuals that live for two periods and leave the economy after period t + 1.

Budget Constraints and Preferences. In the first period of their lives, individuals are unskilled and work in the unskilled sector. In this period of life, individuals make decisions about education, fertility, and savings. Investments in education are denoted by  $e_{t+1} \in [0, 1]$  and increase the level of human capital individuals can supply to the skilled sector in the second period of their lives according to a human capital production function  $h(e_{t+1})$  with h(0) > 0,  $h'(e_{t+1}) > 0$ ,  $h''(e_{t+1}) \leq 0$ , and  $h'(0) < \infty$ .

For simplicity, we restrict attention to consumption during the second period of life. Hence, in the first period of life, individuals decide about how to split their labor income, net of the investment in education  $e_{t+1}$ , between giving birth to children,  $n_t$ , and savings,  $s_t$ . The budget constraint for the first period of life is therefore given by

<sup>&</sup>lt;sup>1</sup>This assumption is standard in the literature, see, e.g., Strulik et al. (2013). The consideration of an extended model with incomplete depreciation is analytically more involved but delivers similar qualitative results; details are available upon request. Moreover, with a realistic depreciation rate of 10% or more per year and a generation length of 20 years, the assumption that physical capital fully depreciates after one generation is quantitatively reasonable.

$$(1 - e_{t+1})w_t^L = n_t + s_t \tag{7}$$

Education imposes (opportunity) costs and the essential trade-off for individuals during their first period of life is that both fertility and education investments imply lower savings.<sup>2</sup>

In the second period of their lives, individuals work in the skilled sector.<sup>3</sup> Consumption is financed from the returns to human capital and the returns to the invested savings from the first period of life, such that the budget constraint in period t + 1 reads

$$(1 + r_{t+1})s_t + w_{t+1}^H h(e_{t+1}) = c_{t+1}$$
(8)

These individual budget constraints hold in every period of life. This implies individuals do not have access to means of transferring resources across periods of life other than education and non-negative savings. In particular, individuals cannot borrow against their future income to finance education or fertility. Combining the period budget constraints (7) and (8) gives the consolidated lifetime budget constraint

$$c_{t+1} = (1 + r_{t+1}) \left[ (1 - e_{t+1}) w_t^L - n_t \right] + w_{t+1}^H h(e_{t+1})$$
(9)

For simplicity, we assume quasi-linear preferences as in Strulik and Weisdorf (2008)

$$U = c_{t+1} + \gamma \ln n_t \quad \text{with } \gamma > 0 \tag{10}$$

where  $c_{t+1}$  denotes consumption and  $n_t$  is the number of children.

**Optimization.** Substituting (9) into (10) allows expressing the lifetime utility maximization as an optimization problem over  $n_t$  and  $e_{t+1}$ , subject to non-negativity constraints. The optimal individual choices are characterized by the following system of first order conditions

$$1 + r_{t+1} = \frac{\gamma}{n_t} \tag{11}$$

$$(1 + r_{t+1})w_t^L \ge w_{t+1}^H h'(e_{t+1})$$
(12)

<sup>&</sup>lt;sup>2</sup>The investment  $e_{t+1}$  could alternatively be interpreted as the share of time spent on education, which reduces effective labor supply to  $1 - e_{t+1}$ . Interpreting the cost of education as resource cost (in terms of resources that are foregone for consumption) is technically convenient as it simplifies the aggregation of labor supply and the analytical derivation of the general equilibrium of the economy, as discussed below. <sup>3</sup>This is consistent with a mild assumption about old individuals having experience that enables them to work in the skilled sector even without spending time on education, and endogenous labor market sorting in equilibrium (as result of a skill premium that is strictly larger than one). This setting is also equivalent

to adopting a vintage human capital perspective where different generations acquire human capital that is specific to operating certain vintages of technology, see, e.g., Cervellati and Sunde (2005).

Notice that an interior solution for fertility is ensured by log-utility. The optimal education choice implies a corner with  $e_{t+1} = 0$  if the return to education is lower than the return on savings,  $(1 + r_{t+1})w_t^L > w_{t+1}^H h'(e_{t+1})$ . In the following we will refer to the corner case of  $e_{t+1} = 0$  as pre-transitional and to the case of  $e_{t+1} > 0$  as post-transitional.

### 2.3 Intra-generational Equilibrium

**Dynamics of Macroeconomic Variables.** Notice that the aggregate stocks of unskilled labor, human capital, and physical capital depend on the fertility, education, and savings decisions of the respective generations. The state variables evolve according to the following laws of motion

$$L_{t+1} = n_t L_t \tag{13}$$

$$H_{t+1} = h(e_{t+1}) L_t \tag{14}$$

$$K_{t+1} = s_t L_t \tag{15}$$

Note that at each point in time the population consists of two generations that are alive, so that the total population size in period t + 1 is given by

$$N_{t+1} = L_t + L_{t+1} = (1+n_t)L_t \tag{16}$$

**Equilibrium.** The general equilibrium requires mutual consistency between optimal individual choices in terms of education, fertility, savings, and the resulting aggregate allocation with the corresponding prices. In the pre-transitional environment with no investment in education (i.e.  $e_{t+1} = 0$ ) optimal fertility is given by the first order condition in (11), that is

$$n_{t} = \frac{\gamma}{1 + r_{t+1}} = \frac{\gamma}{1 - \beta} \frac{s_{t}^{\beta}}{h(0)^{\beta} \left(A_{t+1}^{S}\right)^{1-\beta}} \equiv n\left(s_{t}, A_{t+1}^{S}\right)$$
(17)

where the last equality follows from (6), (14), and (15). Note that fertility is an increasing, strictly concave function of savings and a decreasing, strictly convex function of the level of skilled productivity. As there are no investments in education and hence (12) does not bind, the level of savings is implicitly given by combining the budget constraint (7) with (17)

$$w_t^L = n\left(s_t, A_{t+1}^S\right) + s_t \tag{18}$$

**Lemma 1.** For given, positive levels of  $w_t^L$  and  $A_{t+1}^S$ , there exists a unique level of savings  $s_t = s(w_t^L, A_{t+1}^S)$  with savings being an increasing function of both unskilled wages and

#### skilled productivity.

*Proof.* Note that the right-hand side of (18) is strictly monotonically increasing and concave in  $s_t$ , takes value 0 if  $s_t \to 0$  and converges to infinity for  $s_t \to \infty$ . Hence, the first result follows from the Intermediate Value Theorem. The second result follows from the Implicit Function Theorem as an increase in  $w_t^L$  leads to a monotonic increase in the left-hand side of (18), whereas an increase in  $A_{t+1}^S$  leads to a monotonic decrease in the right hand side. Maintaining equality thus requires  $s_t$  to increase in both cases.

In the post-transitional environment, with positive investment in education, (12) binds with equality. Substituting for factor rents and solving (12) for  $s_t$  gives

$$s_{t} = \frac{1 - \beta}{\beta} \frac{h(e_{t+1})}{h'(e_{t+1})} w_{t}^{L}$$
(19)

This implies that, for any given positive level of education, savings are proportional to wages in the unskilled sector since increasing labor income during the first period of life facilitates savings via an income effect. It is worth noting, however, that once accounting for factor rents and solving for the general equilibrium, savings are increasing in education for any non-convex human capital production function.<sup>4</sup>

Combining the budget constraint (7) and optimal savings (19) yields

$$n_{t} = \left[ (1 - e_{t+1}) - \frac{1 - \beta}{\beta} \frac{h(e_{t+1})}{h'(e_{t+1})} \right] w_{t}^{L}$$
(20)

Optimal fertility is increasing in the level of unskilled wages  $w_t^L$  due to an income effect. Likewise, fertility is decreasing in education for any non-convex human capital production function due to a substitution effect since education imposes an opportunity cost on fertility.

Finally, optimal education is implicitly characterized by combining (11), (12), and inserting (20)

$$\beta(1 - e_{t+1})h'(e_{t+1}) - (1 - \beta)h(e_{t+1}) - \frac{\beta\gamma}{w_{t+1}^H} = 0$$
(21)

Note that this implies an upper bound on education  $e_{max} < 1$  as the left-hand side of (21) becomes negative for  $e_{t+1} = 1$ . Assuming that investments in education are sufficiently effective in producing human capital, such that

$$\beta h'(0) - (1 - \beta)h(0) > 0$$
 (A1)

<sup>&</sup>lt;sup>4</sup>Non-convexity of the human capital production function is a sufficient condition. In fact, it can be easily verified that savings are increasing in education as long as the human capital production function is not too convex, i.e. as long as  $h'(e_{t+1}) h'(e_{t+1}) - h''(e_{t+1}) h(e_{t+1}) > 0$ .

ensures that for some level of  $w_{t+1}^H > 0$  the optimal level of education is larger than zero (i.e. investment in education is profitable in a sufficiently advanced environment). This leads to the following result.

**Lemma 2.** If Assumption (A1) is satisfied, then there exists a level  $\tilde{w}^H > 0$  such that the optimal level of education is given by

$$e_{t+1} = \begin{cases} 0 & \text{if } w_{t+1}^H \le \tilde{w}^H \\ \in (0, \ e_{max}) & \text{if } w_{t+1}^H > \tilde{w}^H \end{cases}$$

and  $e_{t+1} > 0$  is an increasing function of  $w_{t+1}^H$  for any  $w_{t+1}^H \in (\tilde{w}^H, \infty)$ .

*Proof.* Note that (21) is monotonically increasing in  $w_{t+1}^H$ . Moreover, there exists an upper bound  $e_{max}$  such that for  $e_{t+1} > e_{max}$  (21) cannot hold with equality.<sup>5</sup> Hence, if (A1) holds, it follows from the Intermediate Value Theorem that there exits a  $\tilde{w}^H$  such that if  $w_{t+1}^H$  exceeds this value, (21) will hold with  $e_{t+1} > 0$ . As (21) is monotonically increasing in  $w_{t+1}^H$  and monotonically decreasing in  $e_{t+1}$  for any admissible mapping of education into human capital it follows that  $e_{t+1}$  is an increasing function of  $w_{t+1}^H$  if  $w_{t+1}^H \in (\tilde{w}^H, \infty)$ .

#### 2.4The Dynamics of Long-Run Growth

Dynamics of Income and Productivity. To derive the evolution of income per capita over time, consider the dynamics of the different components. The growth of production in the unskilled sector,  $g_{t+1}^{Y_U}$ , depends on the dynamics of the stock of unskilled workers and on the growth of productivity in the unskilled sector

$$1 + g_{t+1}^{Y_U} = n_t^{\alpha} (1 + g_{t+1}^{A_U})^{1-\alpha}$$
(22)

where  $g_{t+1}^{A_U}$  denotes the growth rate of productivity in the unskilled sector.<sup>6</sup>

Analogously, the growth of production in the skilled sector,  $g_{t+1}^{Y_S}$ , depends on the growth of the stock of individual human capital,  $g_{t+1}^h$ , the growth of productivity in the skilled sector,  $g_{t+1}^{A_S}$ , and on the accumulation of physical capital, which depends on the growth of savings,  $g_t^s$ . Substituting the respective terms gives

$$1 + g_{t+1}^{Y_S} = n_{t-1} (1 + g_{t+1}^h)^\beta (1 + g_{t+1}^{A_S})^{1-\beta} (1 + g_t^s)^{1-\beta}$$
(23)

Finally, population growth is given by

<sup>&</sup>lt;sup>5</sup>It is straightforward to show that, as  $w_{t+1}^H \to \infty$ , the upper bound is characterized implicitly by  $\frac{\beta}{1-\beta} \frac{(1-e_{max})h'(e_{max})}{h(e_{max})} = 1.$ <sup>6</sup>See Appendix A.1 for the derivation of this expression as well as of the expressions that follow.

$$1 + g_{t+1}^N = \frac{n_{t-1}(1+n_t)}{1+n_{t-1}}$$
(24)

The dynamics of income per capita can be derived by combining these elements. Let  $\theta_t$  denote the share of the skilled sector in total production,  $\theta_t \equiv \frac{Y_t^S}{Y_t} \in (0, 1)$ . Then using (1) and (24), the growth rate of income per capita can be expressed as

$$1 + g_{t+1}^y = \frac{(1 + g_{t+1}^{Y_U})(1 - \theta_t) + (1 + g_{t+1}^{Y_S})\theta_t}{1 + g_{t+1}^N}$$
(25)

where  $g_{t+1}^{Y_U}$ ,  $g_{t+1}^{Y_S}$  denote the growth rates of total production in the unskilled sector and the skilled sector, respectively.

The evolution of technology is reflected by the productivity dynamics in the two sectors that affect the state variables  $A_t^U$  and  $A_t^S$ . The qualitative predictions of the model do not depend on any specific functional form. To illustrate the working of the model with a simple and transparent benchmark, we derive the analytical results while restricting attention to the case of exogenous productivity growth. In particular, assume

$$A_{t+1}^U = \phi_U A_t^U \tag{26}$$

$$A_{t+1}^S = \phi_S A_t^S \tag{27}$$

with  $\{\phi_U; \phi_S\} > 1$ .

In order to account for the endogenous acceleration of productivity growth and match the model quantitatively to the data, we consider a more realistic formulation that reflects human capital-driven endogenous growth mechanisms in reduced form in the quantitative implementation of the model in Section 3 below.

The Phases of Development. The dynamic evolution of the economy exhibits different phases. Initially, the economy is characterized by a Malthusian phase, which is followed by an endogenous phase transition that is associated with positive and increasing investments in human capital, physical capital, a corresponding fertility transition, and eventually a phase of sustained growth. In the following, we provide an analytical characterization of these phases and of the dynamic evolution of the main variables of interest.

Malthusian Phase. We refer to the Malthusian phase as an environment that is characterized by a setting in which education investments are not profitable and hence  $e_{t+1} = 0$ , as described in Lemma 2. This setting arises as an equilibrium in a technologically underdeveloped economy, in which the level of skilled wages is sufficiently low. Fertility and savings are then characterized by

$$n_t = n\left(s_t, A_{t+1}^S\right)$$
 and  $s_t = s\left(w_t^L, A_{t+1}^S\right)$ 

as implied by expressions (17) and (18). Considering fertility in t + 1, using (6), (14), (15), and re-arranging yields the dynamics of fertility as

$$\frac{n_{t+1}}{n_t} = \left(\frac{s_{t+1}}{s_t}\right)^{\beta} \frac{1}{(1+g_{t+2}^{A_S})^{1-\beta}}$$
(28)

This implies faster fertility growth in phases of faster savings growth and vice versa, but with savings growth exceeding fertility growth. Suppose fertility is above replacement but constant, such that  $\frac{n_{t+1}}{n_t} = 1$ . Since  $1 + g_{t+2}^{A_S} = \phi_S > 1$ , this implies that  $s_{t+1} > s_t$  so that savings increase over time. Moreover, from (18), it follows that in this case the dynamics of savings depend on the dynamics of the unskilled wage. Due to constant growth in productivity in the unskilled sector,  $\phi_U > 1$ , wage dynamics (and hence savings dynamics) depend on the relative sizes of fertility and productivity growth.<sup>7</sup> As an immediate result, the size of the population and the aggregate physical capital stock grow during the Malthusian phase. A noteworthy feature of the model is related to the evolution of income per capita. Although individual human capital remains constant in this phase, productivity grows exogenously and the aggregate stock of capital grows as consequence of individual savings and population growth. This implies that growth of income per capita can be positive during this phase. In Appendix A.2, we derive the lower bound for fertility in the Malthusian phase as

$$n_t > \bar{n}_M = \frac{\phi_U}{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}} \tag{29}$$

Likewise, we show that the upper bound for output per capita growth in the Malthusian phase is given by

$$1 + g_{t+1}^y < 1 + g_{t+1}^{y_M} = \phi_S^{\frac{1-\beta}{\beta}}$$
(30)

**Demographic Transition and Growth Take-off.** The increase in the returns to skills that is due to the growth of productivity and savings implies that the skilled wage  $w_{t+1}^H$ increases monotonically and eventually becomes larger than  $\tilde{w}^H$ . Lemma 2 implies that from this point on, individuals invest in education, such that  $e_{t+1} > 0$ . Since a positive investment in education reduces disposable income, this change in education behavior leads to a reduction in fertility, which constitutes the starting point of the demographic transition. Note that education and savings are complements while education (as well as its complement savings) and fertility are substitutes as shown by (19) and (20). That is, fertility decreases with further expansion of education while savings increase.

 $<sup>^{7}</sup>$ See Appendix A.2 for a formal derivation and for derivations of the following expressions.

As consequence of these behavioral changes, the demographic transition marks a crucial turning point for the dynamics of all state variables of the model. First, the drop in fertility increases the growth rate of per capita income in the unskilled sector. As a result, wages in the unskilled sector increase. Second, this and the increasing education investments have a positive effect on savings. The reason for this is the complementarity between human capital and physical capital in aggregate production, which implies that investments in human capital also increase the returns to savings and vice versa. Together, these effects imply a reallocation of individual expenditures from raising children to investments in human capital and physical capital during the first period of life. A third, more subtle implication is that increases in productivity in the unskilled sector, and hence unskilled wages, reinforce the accumulation of human capital through a positive effect on the skilled wage. This can be seen from substituting (14), (15), and (19) into (5), which yields

$$w_{t+1}^{H} = \beta \left( \frac{A_{t+1}^{S} \frac{1-\beta}{\beta} w_{t}^{L}}{h'(e_{t+1})} \right)^{1-\beta}$$
(31)

Due to the capital-skill complementarity in the skilled sector, the wage in the skilled sector  $w_{t+1}^H$  increases with physical capital. The supply of capital, in turn, increases with the level of unskilled wages from (19). This feature of the model is also quantitatively relevant since it implies that even with constant exogenous technological change in both sectors, skilled wages and investment in education are fuelled by productivity gains in the unskilled sector through the accumulation of physical capital. This effect receives additional momentum after the onset of the demographic transition, with the consequence of a growth acceleration due to factor accumulation.

Subsequently, the demographic transition leads to an acceleration of income growth via the decline in population growth and the associated acceleration in the accumulation of human capital and physical capital. To see this, note that the growth rate in income per capita is still given by (25). However, after the end of the Malthusian phase and with declining fertility, productivity growth now further outweighs the growth in population and hence in the unskilled labor force, that is

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} = \frac{n_t^{\alpha}(1+g_{t+1}^{A_U})^{1-\alpha}}{1+g_{t+1}^N} > \phi_S^{\frac{1-\beta}{\beta}}$$
(32)

implying that the unskilled sector exhibits faster growth in per capita terms. Likewise, the accumulation of human capital (i.e.,  $g_{t+1}^h > 0$ ), and the decline in population growth (i.e., a falling  $g_{t+1}^N$ ) imply an acceleration of growth in per capita terms in the skilled sector. This also implies faster growth in per capita terms than during the Malthusian phase as

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = \frac{n_{t-1}^{1-(1-\alpha)(1-\beta)} \frac{1+g_{t+1}^h}{(1+g_{t+1}^{h'})^{1-\beta}} (1+g_{t+1}^{A_S})^{1-\beta} (1+g_t^{A_U})^{(1-\alpha)(1-\beta)}}{1+g_{t+1}^N} > \phi_S^{\frac{1-\beta}{\beta}}$$
(33)

Growth in income per capita is given by the sum of (32) and (33), weighted by the respective sector shares as in (25). Note that the share of the skilled sector in total production increases over time as the skilled sector grows at a faster rate than the unskilled sector due to factor accumulation.

Another insight from (31) is that the wages in the skilled sector are non-decreasing in the aggregate level of human capital for any non-convex human capital production. This provides a rationale for the observation that a massive increase in human capital does not necessarily imply a reduction in the skill premium, even without skill-biased technical change.

Modern Growth and the Balanced Growth Path. The demographic transition triggers an acceleration of economic development. Ultimately, the economy converges back to a balanced growth path that is based on sustained productivity improvements. To see this, notice that in the long run, education approaches its upper bound, with  $e_{t+1} \rightarrow e_{max}$  as  $t \rightarrow \infty$ . The dynamic system admits a balanced growth path if population grows at a constant rate. Inserting (21) into (20) gives the following condition for a balanced growth path with constant population growth,

$$\frac{n_{t+1}}{n_t} = 1 = \left(\frac{1}{1+g_{t+2}^{A_S}}\right)^{1-\beta} \left(\frac{1+g_{t+1}^{A_U}}{n_t}\right)^{\beta(1-\alpha)} (1+g_{t+2}^{h'})^{\beta}$$
(34)

Inserting the respective growth rates of productivity in the unskilled sector and in the skilled sector, and solving for  $n_t$  given  $\{e_{t+1}; e_{t+2}\} \rightarrow e_{max}$  yields (asymptotic) fertility along the balanced growth path as

$$\bar{n}_{BGP} \to \frac{1 + g_{t+1}^{A_U}}{(1 + g_{t+2}^{A_S})^{\frac{1-\beta}{\beta(1-\alpha)}}} = \frac{\phi_U}{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}}$$
(35)

This balanced growth path exhibits a stable population (see, e.g., Preston et al., 2001, ch. 7) that is characterized by constant birth rates, constant death rates, constant population growth, and a constant age composition. Under a stable population along the balanced growth path, the (asymptotic) growth rate of income per capita is given by evaluating (25) at  $1 + g_{t+1}^h \rightarrow 1$ ,  $1 + g_{t+1}^{h'} \rightarrow 1$ ,  $n_{t-1} = n_t = \bar{n}_{BGP} \rightarrow \frac{\phi_U}{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}}$ ,  $1 + g_t^{A_U} = 1 + g_{t+1}^{A_U} = \phi_U$ , and  $1 + g_{t+1}^{A_S} = \phi_S$ , which yields

$$1 + g_{t+1}^{y_{BGP}} \to \phi_S^{\frac{1-\beta}{\beta}} \tag{36}$$

The growth rate of income per capita on the balanced growth path is lower than during the transition, but higher than during the Malthusian phase. This result is the consequence of the accumulation dynamics of population and human capital, which fade out in the long run. The following proposition provides a summary of the model dynamics.

**Proposition 1.** The development path of the economy exhibits three distinct regimes.

- 1. A Malthusian phase as long as  $w_{t+1}^H < \tilde{w}^H$ , in which the economy is characterized by no investment in education, as well as positive growth of the population, of the aggregate capital stock, and of output per capita.
- 2. An economic and demographic transition, which is characterised by a rapid accumulation of human capital and physical capital, a decline in fertility and population growth, and an acceleration of income per capita growth.
- 3. A phase of modern growth with a balanced growth path that exhibits constant fertility, a stable population with constant population growth, positive human capital investments, and a positive constant growth rate of capital per capita and income per capita. Income per capita grows at a faster rate than during the Malthusian phase, but at a lower rate than during the transition.

*Proof.* The statements follow immediately from Lemma 2, (17)-(21), as well as (30), (32), (33), (35), and (36).

### 2.5 Extensions

The model developed so far describes the dynamics of a single country in autarky with exogenous productivity growth in both sectors. The purpose of the model is to illustrate the role of the demographic transition for the development dynamics of a country and the resulting empirical implications. The assumptions that productivity growth is purely exogenous and that all countries undergo the same dynamic process are restrictive and counterfactual. In the quantitative analysis of the model, we relax both assumptions, one at a time, to obtain a more realistic representation.

Human Capital Externalities. While the baseline model with purely exogenous productivity growth is able to account for the qualitative patterns of income per capita, in particular for a temporary acceleration of growth in the context of the economic and demographic transition, this model is unable to match the quantitative acceleration in productivity and income per capita growth after the transition. It is well established that factor accumulation alone cannot explain the divergence in incomes (see, e.g., Hsieh and Klenow, 2010, and Sunde et al., 2021). In order to match the quantitative patterns of income per capita, we consider an extension of the model with endogenous technological change. In particular, we consider a reduced-form representation that allows for human capital externalities on productivity growth in the form of a "level effect", according to which a higher level of human capital leads to higher rates of productivity growth, and in the form of a "growth effect" of human capital on productivity growth, according to which a faster expansion of human capital implies a faster rate of productivity growth.

**Technology Diffusion.** Some of the facts of long-run growth explicitly refer to comparative development patterns and cross-country comparisons. To account for these patterns, we consider a multi-country version of the model with international diffusion of technology, which allows us to account for the possibility of cross-country technology spill-overs and assess their relevance in accounting for the facts. Specifically, we consider an extension of the model with technology diffusion in the spirit of Nelson and Phelps (1966). This extension does not affect the timing of the demographic transition, since technology imports are not possible without education. However, it clearly affects the equilibrium dynamics after the onset of the demographic transition by accelerating technological development and, as consequence, subsequent economic and demographic development. At the same time, this helps concentrating on the central role of the demographic transition by allowing only one channel through which international spillovers affect the development trajectory of a country.<sup>8</sup> As will be shown in more detail below, this extension provides novel empirical implications. In particular, countries that undergo the demographic and economic transition later have access to technologies developed in more developed countries that experienced an earlier transition.

### 2.6 Parametrization

We end the discussion by laying out a quantitative version of the baseline version of the model. In the following, we calibrate the model to match empirical data moments for England. In the next section, we will compare the simulated data to actual statistical data and investigate whether the model can serve as a data generating process that accounts for the empirical patterns of long-run growth.

The simulation makes use of the fact that, given a set of initial values for  $L_t$ ,  $H_t$ ,  $K_t$ ,  $A_t^U$ , and  $A_t^S$ , the productivity levels  $A_{t+1}^U$  and  $A_{t+1}^S$  are determined. This allows computing the equilibrium allocation in terms of  $n_t$ ,  $s_t$ , and  $e_{t+1}$ , and thus  $L_{t+1}$ ,  $K_{t+1}$ , and  $H_{t+1}$ , yielding the state variables that are needed to simulate the entire development

<sup>&</sup>lt;sup>8</sup>Alternative dimensions for international spillovers are trade in goods and international capital mobility. As argued by Comin and Mestieri (2018), models of technology diffusion are more suitable for studying income divergence than trade-based theories, which are inconsistent with the divergence in incomes across countries that was observed between 1913 and the 1970s despite the collapse in world trade between those points in time. Similarly, international capital mobility was historically subject to various degrees of restrictions, adding complications that go beyond the purpose of this paper.

path forward.

**Data.** The data moments that serve as targets for the model calibration are data for England from the Maddison data base (Bolt and van Zanden, 2020). We also make use of data for the timing of the demographic transition by Reher (2004), data for education by Lee and Lee (2016), and data for comparative development from the Penn World Tables (Feenstra et al., 2015, version 9.1).

**Calibration.** We set the length of a generation to be 20 years and simulate the model for a sequence of 35 generations beginning with the year 1700, which illustrates the working of the model over the long run. The choice of the simulation window is without consequence for the main results and serves the purpose of illustration. Below we will restrict the simulated data to a time frame that is comparable to the observed panel data in some of the empirical analysis. It is important to note, however, that all simulations are based on the same long simulation window.<sup>9</sup>

For simplicity, we consider a parameter-free, linear production function of human capital with  $h(e_{t+1}) = 1 + e_{t+1}$ . This implies an optimal education level of  $e_{t+1} = \max\left(2\beta - 1 - \frac{\beta\gamma}{w_{t+1}^H}, 0\right)$  as well as  $h'(e_{t+1}) = 1$ , h(0) = 1, h'(0) = 1, and  $e_{max} = 2\beta - 1$ .

The parameters of the model are set to match data moments for England (United Kingdom) in the Maddison database by Bolt and van Zanden (2020). The parameter X (land size) plays no crucial role except for determining initial values and therefore can be normalized to one. We set the capital income share in the skilled sector to  $\frac{1}{3}$ , which implies setting  $\beta = \frac{2}{3}$ , in line with standard calibrations of growth or business cycle models. The labor share in the unskilled sector is set to  $\frac{1}{2}$ , which implies a value of  $\alpha = 0.5$ , similar to values used by Hansen and Prescott (2002) and Doepke (2004).

Assuming an education level of zero as well as constant fertility n, iterating (16) forward yields population size at time t > 0 as

$$N_t = (1+n) n^{t-1} \left( N_0 - \frac{N_0}{1+n} \right)$$

where the initial population level  $N_0$  is normalised to one. Assuming the demographic transition to occur in 1880, which corresponds to a take-off in education from 1880 to 1900 (in line with Lee and Lee, 2016), we obtain the population level just before the onset of the demographic transition from the Maddison database and normalise it by the size of the population in the year 1700 (i.e., initial population size). This allows computing the average fertility n that matches the two values. As fertility in the model

<sup>&</sup>lt;sup>9</sup>The choice of the simulation window is driven by considerations concerning the availability of comparable and reliable data for the period before 1700. The qualitative and quantitative results of the simulation are not sensitive to the size of the simulation window. The model can be simulated for an arbitrarily long period of (quasi-)stagnation by adjusting the respective parameters accordingly.

is increasing during this phase, n constitutes an upper bound and we set the target  $\bar{n}_M$  to the arithmetic mean between n and replacement fertility. This yields the initial stocks of aggregate unskilled labor  $L_0 = N_0 - \frac{N_0}{1+\bar{n}_M}$  and aggregate human capital  $H_0 = h_0 \frac{N_0}{1+\bar{n}_M}$ . Moreover, assuming constant growth of GDP per capita gives GDP per capita at time t > 0 as

$$y_t = (1+g)^t y_0$$

where again the initial level of GDP per capita  $y_0$  is normalised to one. We retrieve the level of GDP per capita just before the onset of the demographic transition from the Maddison database and normalise it by GDP per capita in the year 1700. Hence we can compute the average growth rate g that matches the two values. As GDP per capita growth varies during this period we set our target  $g^{Y_M}$  to the arithmetic mean between gand 0.4 percent per annum, which is consistent with empirical estimates for the 18<sup>th</sup> and early 19<sup>th</sup> century (see, e.g., Crafts, 2021). Since (approximate) GDP per capita growth in the Malthusian phase is given by (30) this allows computing  $\phi_S \approx 1.189$ . Given  $\phi_S$  and (approximate) fertility given by  $\bar{n}_M$  as in (29), we can compute  $\phi_U \approx 1.2866$ .

In order to compute the initial productivity level of the unskilled sector, we set the initial share of income devoted to children  $\varepsilon$  to approximately two thirds, which implies an initial savings level of  $s_0 = (1 - \varepsilon) w_0^{L,10}$  The savings level, together with initial skilled productivity, determines the initial skilled wage and hence the timing of the demographic transition. The initial level of productivity in the unskilled sector can then be obtained by solving (7) for  $e_{t+1} = 0$  and  $n_0 = \bar{n}_M$ 

$$w_0^L = \bar{n}_M + s_0 \iff A_0^U = \frac{L_0}{X} \left(\frac{\bar{n}_M}{\varepsilon}\right)^{\frac{1}{1-\alpha}} \approx 1.3277$$

As consequence, this yields the initial capital stock as  $K_0 = \frac{s_0}{1+g^{Y_M}} \frac{N_0}{1+\bar{n}_M} \approx 0.2274$ . The initial condition for productivity in the skilled sector is set to satisfy  $w_t^H(1+e_t) \geq w_t^L$   $\forall t$ , i.e., a skill premium that always exceeds one. A sufficient condition for this to hold is given by

$$A_0^S \ge \frac{H_0}{K_0} \left(\frac{w_0^L}{\beta h_0}\right)^{\frac{1}{1-\beta}} \approx 29.1066$$

which also delivers a demographic transition in 1880. The parameter  $\gamma$  is set to ensure  $n_0 = \bar{n}_M$ . Given  $\beta$ ,  $s_0$ ,  $h_0$ ,  $\phi_S$ ,  $A_0^S$ , and  $\bar{n}_M$ , solving (17) for  $\gamma$  yields  $\gamma \approx 1.8265$ .

To account for the endogenous acceleration of productivity growth, we consider a formulation that contains an exogenous component and includes an endogenous growth component based on human capital externalities. More specifically, as extension to (26)

<sup>&</sup>lt;sup>10</sup>This corresponds to spending approximately 50 percent of family income on children, in line with evidence by Turvey (2010) who points towards a yearly expenditure of around  $8\pounds$  per child in 1750.

 Table 1: Calibration: Baseline Model

Parameter	Value	Target
X	1	normalised
$\beta$	$\frac{2}{3}$	capital share of $\frac{1}{3}$
$\alpha$	$0.\check{5}$	labor share of $\frac{1}{2}$
$N_0$	1	normalised
$\phi_S$	1.189014	(relative) GDP per capita of England in 1860 (from Maddison)
$\phi_U$	1.286587	(relative) population level of England in 1860 (from Maddison)
$A_0^U$	1.327654	initial expenditure share of fertility of approximately $\frac{2}{3}$
$A_0^S$	29.10659	skill premium
$\gamma$	1.826472	initial fertility level of $n_0 = \bar{n}_M$
$\sigma_S$	0.756549	BGP growth rate of $1\%$ p.a. (from Maddison)
$\sigma_U$	0.4716599	replacement fertility along the BGP
$\psi$	11.66559	peak growth rate of $2.25\%$ p.a. (from Maddison)

and (27), we consider

$$A_{t+1}^{U} = \phi_{U} \left\{ 1 + \sigma_{U} \left[ h\left( e_{t+1} \right) - h_{min} \right] + \psi \left[ h\left( e_{t+1} \right) - h\left( e_{t} \right) \right] \right\} A_{t}^{U}$$
(37)

$$A_{t+1}^{S} = \phi_{S} \left\{ 1 + \sigma_{S} \left[ h\left( e_{t} \right) - h_{min} \right] + \psi \left[ h\left( e_{t} \right) - h\left( e_{t-1} \right) \right] \right\} A_{t}^{S}$$
(38)

Table 1 summarizes the values and targets of the baseline calibration.

Several aspects of this formulation are worth noting. First, with the linear human capital production function,  $h_{min} = h(0) = 1$ . This implies that this formulation does not affect the rest of the calibration, particularly the timing of the demographic transition, as the additional parameters  $\sigma_U$ ,  $\sigma_S$ , and  $\psi$  have no effect when  $e_{t+1} = 0$ . Second, the terms  $\sigma_U [h(e_{t+1}) - h_{min}]$  and  $\sigma_S [h(e_t) - h_{min}]$  account for the "level effect" of human capital on productivity growth, i.e., an expansion of human capital leads to higher rates of productivity growth, ceteris paribus. Analogously, the terms  $\psi [h(e_{t+1}) - h(e_t)]$ and  $\psi [h(e_t) - h(e_{t-1})]$  account for the "growth effect" of human capital on productivity growth, that is, productivity growth is more rapid when the stock of human capital expands. Existing evidence suggests that both channels are empirically relevant, such that  $\{\sigma_U, \sigma_S, \psi\} > 0$  (see, e.g., Sunde and Vischer, 2015). Third, the timing of this formulation implies that a generation does not exhibit an externality on itself, as future skilled productivity does not depend on the education choice of the current (unskilled) generation.

The parameter  $\sigma_S$  is calibrated to generate a balanced growth path growth rate of income per capita of one percent per year, which is now given by

$$1 + g_{t+1}^{y_{BGP}} \to \left[\phi_S(1 + \sigma_S e_{max})\right]^{\frac{1-\beta}{\beta}}$$

Given  $e_{max} = 2\beta - 1$  this yields  $\sigma_S \approx 0.7565$ . The parameter  $\sigma_U$  is calibrated to generate an (asymptotic) fertility level along the balanced growth path at replacement (and hence a stationary population size). Solving (35) for  $\bar{n}_{BGP} = 1$  yields  $\sigma_U \approx 0.4717$ . Lastly, the parameter  $\psi$  is calibrated to generate a peak growth rate of income per capita of 2.25 percent per year. Given all other parameter values, solving for the peak growth rate of income per capita to equal 2.25 percent per year gives  $\psi \approx 11.6656$ .

## 3 Accounting for Long-Run Growth Dynamics

We are now in a position to assess whether the model developed in the previous section is able to match the stylized empirical facts of long-run growth. This section considers the implications of the baseline model, focusing on the behavior of a single country in isolation. The next section assesses whether the cross-country version of the model can account for the patterns of comparative development across countries. Our exposition follows the six "New Kaldor Facts" proposed by Jones and Romer (2010) and provides an updated and extended version that considers additional evidence that has become available more recently and that led to a more nuanced view of the facts of long-run growth.

### 3.1 Simulated Long-Run Development Dynamics

We begin the analysis by simulating the quantitative version of the model before contrasting it with empirical patterns in various dimensions. The main purpose of this analysis is to illustrate the main features of the model, including the non-linear long-run dynamics in the core variables, namely output, population, physical capital, and human capital.

Figure 1 Panel (a) plots the long-run evolution of (annualized) growth rates of population and output per capita for the baseline country (England) over the period 1700 to 2100, while Panel (b) plots the evolution of all state variables in addition to output per capita growth. The simulation pinpoints several characteristic features of the unified growth model. As consequence of the choice of parameters and initial conditions, the model exhibits a demographic transition with an onset of the decline in fertility in 1880. Population grows before this point, while the growth rate declines thereafter and converges to a stable population with fertility at replacement.

Panel (b) illustrates the dynamics of the state variables population, physical capital, education, and productivity. The fertility transition marks a turning point in the model dynamics, as declining fertility gives rise to increased education attainment per capita. As consequence of productivity growth, output per capita exhibits a small but positive growth rate even before the demographic transition. It is, however, the demographic transition and the associated co-evolution of human capital accumulation and the intensified



(a) Population and Income Dynamics (b) Dynamics of State Variables

Figure 1: Simulated Development Path of the Baseline Economy

accumulation of physical capital that leads to a sharp acceleration in output per capita growth. This acceleration is strengthened by the aggregate complementarity of physical capital and human capital. Growth in output per capita also benefits from the reduction in population growth and from the positive effects of the education expansion on productivity growth. Eventually, however, this process loses momentum and growth peaks in the later phases of the demographic transition before it converges to a lower sustained growth path thereafter. Although output per capita growth remains permanently higher than before the demographic transition, it declines once the transition dynamics lose momentum. As consequence, aggregate output per capita growth stabilises at a growth rate that is higher than during the Malthusian phase, but lower than during the transition when output per capita growth is affected by the one-off dynamics related to the demographic transition.

## 3.2 Accounting for Empirical Growth Dynamics

### 3.2.1 The Secular Acceleration of Growth

The most salient empirical fact of long-run growth refers to the secular acceleration of development. Conventionally, the stylized presentation of this fact juxtaposes a phase of stagnant income per capita with a sharp acceleration to sustained income growth in the Western world (see, e.g., data from the Maddison Project, Bolt et al., 2018, 2020). The inconsistency of this observation with models of balanced growth sparked the development of theories that explain the income acceleration exclusively by the accumulation of physical capital (Hansen and Prescott, 2002), or by theories that capture also other stylized facts related to the concurrent demographic transition in fertility from a Malthusian regime to a modern regime with fewer but better educated children (see, e.g., Galor and Weil, 2000, Galor, 2011).

As a unified growth framework, the model is consistent with the secular acceleration from a quasi-stagnant environment to sustained growth. The simulation results presented above illustrate that the model exhibits a long-run development path that is initially characterized by a Malthusian phase that is associated with slow (but non-zero) growth in income per capita, high fertility, and no education. The last phase of development is characterized by a balanced growth path with low fertility and a positive level of education. The intermediate phase is given by a demographic and economic transition, during which growth of income per capita accelerates and temporarily exceeds the growth rate of the balanced growth path. Figure 2 documents that the calibrated version of the model closely resembles the long-run growth dynamics of income per capita and population observed for England when abstracting from the fact that stochastic factors that severely affected economic growth, such as World War I, are not accounted for by the model by construction.



Figure 2: Growth Dynamics – England Data: Maddison Project, Bolt and van Zanden (2020).

Five aspects are noteworthy. First, the growth trajectory exhibits a phase transition from an almost stagnant development path to a sustained growth regime. This transition occurs over less than two centuries and marks the end of Malthusian dynamics and the beginning of a modern growth regime.

Second, the acceleration of income growth eventually fades out and growth rates stabilize at a level below their intermediate peak. The reason for this is that the education expansion eventually ends as education converges to its endogenous upper bound, as discussed in more detail below. As a consequence, the acceleration in growth due to capital-skill complementarities and the acceleration of technological progress due to human capital also loses momentum. This feature is consistent with the observation of a "wave pattern" in productivity growth that occurs in the context of the education expansion (see, e.g., Rangazas, 2002).

Third, the growth acceleration leads to a temporary overshooting of income growth above the endogenous growth rate that is sustainable in the long run. This pattern is consistent with an empirical aspect of long-run growth that has received less attention than the take-off. The slow-down of income dynamics after the growth take-off has implications for the recent debate about "secular stagnation", which mostly focused on the income growth slowdown observed in developed countries over the last decades. In search of the causes for this slowdown, some scholars raised concerns about the dynamism of technological progress and demographic factors that impose increasing impediments for development (see, e.g., Teulings and Baldwin, 2014, OECD, 2015, Gordon, 2012, 2014, 2016). In light of the finding of a hump-shaped pattern of growth also among low-income countries, some have pointed at the potential role of transition dynamics (see, e.g., Gundlach and Paldam, 2020). In fact, evidence by Cervellati et al. (2017) suggests that part of the slowdown might be explained by the acceleration of growth after the demographic transition, which fades out as the temporary dynamics of the sudden accumulation of production factors come to an end. The model provides a rationalization for this hypothesis. The overshooting and subsequent slow-down of growth is related to the additional momentum generated by the sudden accumulation of human capital and physical capital, which are strategic complements from the perspective of individuals. Once this rapid accumulation, which is the direct consequence of the singular, non-recurrent demographic transition, fades out, the only source of growth is sustained productivity improvement, and hence the standard mechanism giving rise to balanced growth. These non-monotonic dynamics, which are consistent with the empirical patterns, differ from previous accounts that considered monotonic convergence dynamics to the balanced growth path.

Fourth, the decline in growth during the approach to the last phase of development has implications for the prospects of future growth on a global scale. This suggests that it might be natural to expect slower growth in the future for countries that underwent the demographic transition in the past as result of the non-recurring take-off dynamics in the context of the demographic and economic transition. The reinforcing interactions between population, human capital, physical capital, and technology, first lead to an acceleration of growth but eventually fade out. In this respect, the model contributes a novel aspect to the secular stagnation debate, which has focused on a slow-down in technology and considered the role of demographic factors primarily from a perspective of aging and public health (Gordon, 2012, 2014, 2016).

Fifth, despite a Malthusian population regime, which is reflected in significant population growth the model exhibits very small but sustained positive growth in income per capita prior to the transition. This is in line with recent empirical findings of moderately positive growth in incomes during the 17<sup>th</sup> and 18<sup>th</sup> century that has been viewed as challenging the empirical validity of existing unified growth theories that are based on a Malthusian phase of stagnation (see, e.g., Fouquet and Broadberry, 2015, p. 228) and

Broadberry et al., 2015). More recent evidence suggests that earnings trended upwards since the late 16<sup>th</sup> century, while the increase in incomes was associated with a secular decline in the labor share (Humphries and Weisdorf, 2019). The moderate increase in earnings during this phase of population expansion is consistent with technological progress (see Crafts and Mills, 2021).<sup>11</sup> This modest increase in economic activity prior to the take-off in growth has so far only been detected for Europe. This pattern can be viewed as consequence of Malthusian fluctuations, when reproductive success does not keep pace with productivity increases. In the centuries prior to the demographic transition, improvements in productivity were associated, for instance, with the relaxation of land constraints due to the discovery of the New World, and with gains from trade in the context of increased globalization. Nevertheless, the analysis of the model documents that a unified growth approach is perfectly consistent with evidence of gradually increasing income even before the transition. At the same time, the model analysis also puts this sluggish growth in perspective against the eruptive acceleration of growth during the middle of the 19<sup>th</sup> century, roughly at the same time when physical capital deepening and human capital deepening experienced a significant acceleration, consistent with recent evidence (see, e.g., Crafts, 2021, Tables 2 and 3). Moreover, the model underscores the important quantitative and qualitative difference between non-zero but slow growth prior to the transition, and the eruption of growth and development across all dimensions of the economy that occurs in the context of the demographic transition.

### 3.2.2 Population and Demographic Dynamics

Historically, demographic development was characterized by stable populations and moderate levels of population growth. Eventually, this phase was followed by a demographic transition that brought about a sharp decline in fertility rates. Although this demographic transition occurred simultaneously to the acceleration in income growth this reversal in population growth rates plays no role in early theories explaining the population dynamics (see, e.g., Kremer, 1993) or the acceleration of economic development based on the accumulation of physical capital (see, e.g., Hansen and Prescott, 2002). In contrast, the demographic transition is a central element in explaining the growth acceleration in unified growth models, with the decline in fertility reflecting the onset of investments in education (see, e.g., Galor and Weil, 2000, Galor, 2011). Delays in this transition have been associated with the great divergence in comparative development across the world (see, e.g., Galor, 2011, or Cervellati and Sunde, 2015).

The demographic transition is usually seen as the end of a Malthusian phase of popula-

<sup>&</sup>lt;sup>11</sup>Similar findings of increasing population and steadily increasing production, mostly due to an increasing share of GDP accounted for by non-agricultural production, have been reported for Italy (Malanima, 2011), Holland (van Zanden and van Leeuwen, 2012), Spain (Alvarez-Nogal and Prados de la Escosura, 2013), Sweden (Schön and Krantz, 2015), and Portugal (Palma and Reis, 2019).

tion development. Malthusian dynamics in the pre-industrial era imply that productivity improvements were absorbed by an increasing population (see, e.g., Ashraf and Galor, 2011). The main reason that greater population eroded improvements in living conditions per capita was related to the presence of fixed factors and the associated decreasing returns to labor. However, related to the previous discussion, recent findings suggest that the population increase was indeed associated with a very moderate improvement in income per capita (see, e.g., Bolt and van Zanden, 2020). This suggests the existence of scale effects, which might operate through agglomeration effects, knowledge spillovers, and ultimately improvements in productivity that trigger the demographic transition (as in, e.g., Galor and Weil, 2000) and foster subsequent growth.

The model is consistent with a rapid population expansion prior to the demographic transition and a convergence to a stable population thereafter. Moreover, as illustrated by Figure 3, the population expansion occurs substantially before the growth acceleration. Again, this figure illustrates the dramatic increase in income per capita over the past two centuries, which dwarfs the economic development in the centuries before.<sup>12</sup>



Figure 3: The Expansion of Income and Population – England Data: Maddison Project, Bolt and van Zanden (2020).

Several aspects are particularly noteworthy in this context. First, the simulated model stipulates a convergence to a balanced growth path with stationary population size. Empirically, the population dynamics in the aftermath of the onset of the demographic transition are less well understood than the dynamics prior to the transition. Most accounts argue that, based on intuition, population growth will eventually fade out (see, e.g., Jones and Romer, 2010). Unified growth models often impose the convergence to a balanced

<sup>&</sup>lt;sup>12</sup>While not accounting for an extension of the market in terms of urbanization or globalization the model is consistent with an association of growth acceleration with scale effects, as suggested by Jones and Romer (2010). Considering the possibility that the unskilled sector reflects the primary (agricultural) sector of production, whereas the skilled sector reflects the secondary (manufacturing) or tertiary (service) sectors, which are predominantly located in urban agglomerations, the model dynamics are consistent with a narrative of urbanization. See also Baudin and Stelter (2022) for recent work on this aspect.

growth path with stationary population (see, e.g., Galor, 2011, or Cervellati and Sunde, 2015). In other accounts, the asymptotic population dynamics are left unspecified (Strulik et al., 2013). Evidence for population dynamics for the last decades documents that fertility often drops below replacement in the process of the demographic transition. While the majority of post-transitional countries display total fertility rates around or below replacement, total fertility has dropped in some cases to levels below 1.5. There is an ongoing discussion whether this lowest-low fertility scenario is a temporary phenomenon (see, e.g., Bongaarts, 2002, Billari and Kohler, 2004, Goldstein et al., 2009, and Myrskylä et al., 2009). In the model simulation, the stabilization of population in the long run is due to the specific calibration, as discussed above. An alternative calibration could account for sustained positive or negative population growth along the endogenous growth path. In the unified growth literature, this is often implicitly assumed or neglected as an asymptotic behavior of little practical relevance for understanding the transition dynamics.

Second, in the baseline calibration, the model exhibits a monotonic decline in population growth during the demographic transition. In general, however, the model allows for non-monotonic population dynamics. As becomes clear from inspecting (34), fertility dynamics depend on an interplay of income and substitution effects, where the income effect is related to the dynamics of the wage (and hence productivity) in the unskilled sector, whereas the substitution effect is related to the dynamics of productivity in the skilled sector, which provides the incentives for education and savings. In the baseline specification of the productivity dynamics in (37) and (38), productivity in the unskilled sector is affected through intra-generational spillovers of human capital (by considering an influence of  $h(e_{t+1})$  on productivity  $A_{t+1}^U$ , while productivity in the skilled sector is affected through inter-generational spillovers (by considering an influence of  $h(e_t)$  on productivity  $A_{t+1}^S$  as in much of the unified growth literature (see, e.g., Cervellati and Sunde, 2005). This implies a delay in the skilled sector and hence a relatively weak substitution effect in the fertility choice. Choosing a symmetric formulation (i.e., where both technologies are affected by intergenerational spill-overs) implies a stronger substitution effect, and hence a sharper decline in fertility during the early phase of the demographic transition. As illustrated in Figure 4, this leads to a temporary decline in fertility below replacement even if the population stabilizes at a stationary level during the phase of modern growth. This provides a rationalization for the observed drop of fertility below replacement in many countries. The calibration of a balanced growth path with stationary population implies a rebound, which sheds new light on the empirical debate about the reversal in fertility dynamics in many of the "lowest-low" fertility countries.



Figure 4: Alternative Scenario: Fertility Reduction Below Replacement Data: Maddison Project, Bolt and van Zanden (2020).

#### 3.2.3 The Expansion of Human Capital and Physical Capital

The key mechanism behind the acceleration of growth in income per capita and the slowdown in population growth is related to the expansion of education and the corresponding decline in fertility in the context of the demographic transition. The massive expansion of human capital over the past two centuries represents another stylized fact of long-run growth (Jones and Romer, 2010, Lee and Lee, 2016). In light of converging evidence that human capital plays an important role for economic development (see, e.g., Gennaioli et al., 2013, Sunde and Vischer, 2015, Hanushek et al., 2017), this expansion is viewed as a major driver of the growth acceleration. While the onset of this expansion in Western countries occurred in the temporal context of the demographic transition (see, e.g., Lee and Lee, 2016), there is an ongoing discussion about whether education was a determinant or consequence of the acceleration in growth. This question also relates to the dynamics of wages and the skill premium, which is discussed below.

Figure 5 illustrates the expansion of education in England over the past two centuries and compares the model dynamics to the empirical patterns. The dynamics in education investments (in terms of  $e_{t+1}$  or a derived measure of years of schooling) captures the takeoff and the overall dynamics in empirical measures of human capital fairly accurately, regardless of the precise measure used for education.<sup>13</sup> In particular, the take-off in education investments captures the take-off in average years of schooling and a human capital index. Considering enrolment rates, the take-off in education investments occurs

<sup>&</sup>lt;sup>13</sup>In the model, education investments are captured by  $e_{t+1}$ . As discussed in the introduction, the model is based on a broad notion of skilled human capital, and a conceptually simple and transparent model of skill acquisition as an individual's choice between savings, own education, and fertility. This modeling strategy is consistent with higher levels of education, such as secondary enrolment. A model of parental education choice would likely deliver similar predictions regarding the nexus of education and fertility in the long-run, but would be more suitable for lower levels of education. To construct a measure of years of schooling, we normalize education with  $e_{max}$  to represent 14 years of schooling.

with a considerable delay relative to primary enrolment, but closely tracks secondary enrolment. While qualitatively similar, the take-off in tertiary enrolment came later.



Figure 5: The Expansion of Human Capital – England Data: Lee and Lee (2016).

A similar expansion has been observed for physical capital. Providing estimates of historical capital stocks has been notoriously difficult, but recent estimates by Broadberry and de Pleijt (2021) indicate that a non-negligible part of output growth during the early phases of the growth acceleration was driven by capital accumulation, which experienced a visible acceleration during the middle of the 19<sup>th</sup> century. Prior to the acceleration, the capital stock exhibited visible increases as early as during the 17<sup>th</sup> century, but the evidence suggests that capital intensity did not increase or even decreased in the context of rapid population expansion. The estimates by Broadberry and de Pleijt (2021) for Great Britain also indicate that the importance of (fixed) capital for production increased while the importance of land declined as early as in the 17<sup>th</sup> century. Their estimates also show that the capital-output ratio declined during this period.

Figure 6 presents the corresponding results for the accumulation of physical capital. Consistent with the discussion of the analytical predictions for the Malthusian phase, individual savings imply an increase in the aggregate capital stock. However, the available resource endowments (in terms of capital per capita and land per capita) slowly decrease as the result of Malthusian dynamics, as illustrated in Panel (a). As soon as the fertility decline begins around 1880, the resource endowments per capita increase in the model. An increase is also visible in the data for the post-transitional period, as illustrated by Panel (b). As consequence of the dynamics of capital endowments per capita and moderate increases in income per capita, the capital output ratio falls during the Malthusian phase. This resembles the decline in the capital-output ratio that has been documented for England for the late 17<sup>th</sup> to the mid 19<sup>th</sup> century (see, e.g., Broadberry and de Pleijt, 2021).



(a) Capital p.c. – Malthusian Phase (b) Capital p.c. – Post-Transition

Figure 6: The Expansion of Physical Capital – England Data: (a) Broadberry and de Pleijt (2021), Figure 4b (fixed capital excl. dwellings, own digitization). (b) Penn World Tables (Feenstra et al., 2015, version 9.1). Model: fixed capital per capita.

#### 3.2.4 The Dynamics of the Skill Premium and Labor Share

An empirical observation that has generated considerable interest is the development of the skill premium. In particular, despite the massive expansion of human capital, there has been no observation of a marked decline in the skill premium. If anything, the skill premium seems to have widened during the 20<sup>th</sup> century, even in developing countries (see, e.g., Goldberg and Pavcnik, 2007, and Jones and Romer, 2010). Candidate explanations for this observation are skill-biased technical change (Goldin and Katz, 2007), or complementarities between human capital and physical capital (as in, e.g., Krusell et al., 2000).

The literature has also pointed out a tension between a human capital-driven acceleration as in unified growth theories and the empirical dynamics of wages and skill premia in the historical context prior and around the time of the growth take-off, mostly for the case of Great Britain. Van Zanden (2009) documented that the skill premium in Western Europe had not increased markedly during the late medieval period and prior to the growth acceleration, while Betrán et al. (2010) show evidence for a decline in the skill premium during the late 19<sup>th</sup> century. Others, including, e.g., Feinstein (1998) and Allen (2009) argued that real wages stagnated while output per worker expanded during the early phases of the economic transition. The explanation for this was seen in technical change increasing the demand for physical capital, which led to an intermediate increase in the capital income share, which was ultimately followed by an increase in wages. Recent evidence by Humphries and Weisdorf (2019) complements this by showing that the labor share exhibited a temporary minimum during the onset of the growth take-off. These observations have raised doubt about the compatibility of the evidence with unified growth models in which an increase in the demand for skilled labor is typically seen as the major driver behind the onset of the demographic transition and the associated increase in education. At the same time, there is evidence that suggests that education was an important factor during all phases of industrialization, (see, e.g., Becker et al., 2011).

As shown before, the model dynamics exhibit a sustained increase in human capital. Due to the complementarity between human capital level and the demand for human capital that results from capital-skill complementarity, the model does not predict a monotonic decline in the skill premium. This is illustrated in Figure 7, which plots the earnings premium and the wage ratio of the skilled sector relative to the unskilled sector for a re-calibrated version of the model in which the education transition begins in 1920 to have a direct comparison with the college and high school premium for the US reported by Goldin and Katz (2007a). While there is a slight difference in the levels, the overall pattern is strikingly similar. In particular, during the early phases of the transition, the skill premium declines. Despite this, the supply of skilled labor continues to increase, as shown in Figure 5, and the skill premium even rebounds during the later phases. While in principle, this is consistent with a race between education and technology as discussed by Goldin and Katz (2007a), the model dynamics are also related to the accumulation of physical capital during this period and the associated feedback loop to education.



(a) College/High School Wage Premium (Data, (b) Skill/Earnings Premium (Model) USA)

#### Figure 7: The Dynamics of the Skill Premium

A related question concerns the dynamics of the income shares of labor and capital. Historically, the capital share exhibited a moderate upward trend. Figure 8 illustrates that the model is consistent with this observation. Panel (a) plots the dynamics of the capital income share for the baseline economy and the corresponding data for England. Underlying the increase in the capital share is a continuing decline in the overall labor income share, which is predominantly due to the decline of the labor income share from the unskilled sector. In fact, the skilled labor share increases despite the decline of the overall labor share. Moreover, this decline exhibits a moderate reversal during the early phase of the demographic transition, consistent with a temporary minimum during the

Data: (a) Goldin and Katz (2007b), Figure 1, Table A8.1 (own digitization). (b) Baseline model, recalibrated for an education transition in 1920 (corresponding to the USA).

onset of the growth take-off in Britain reported by Humphries and Weisdorf (2019). This is the result of general equilibrium effects. Once the education expansion sets in, the labor income share increases temporarily, leading to a temporary decline of the capital share during the early phase of the transition.

Panel (b) of Figure 8 shows the dynamics of the returns to capital implied by the model. The real return exhibits a moderate decline prior to the onset of the transition, and a modest increase thereafter. When accounting for the variation in productivity (by normalizing the dynamics of the real return relative to income dynamics), the real return exhibits a steady decline. These dynamics are consistent with the overall pattern found in the data.<sup>14</sup> Again this is the result of general equilibrium effects that lead to capital deepening. In addition, the model also predicts a secular decline in the return of capital relative to the return on (skilled or unskilled) labor as consequence of continued capital accumulation. This complements arguments related to a savings glut that have been made in the recent secular stagnation debate (see, e.g., Teulings and Baldwin, 2014, and OECD, 2015).



Figure 8: The Dynamics of Capital Shares and Real Returns Data: (a) England, Crafts (2021), Table 10. (b) Schmelzing (2020), Figure IV (global rate, GDP-weighted). Simulated data: (a)  $(R \cdot K)/(Y^U + Y^S)$ . (b)  $r = (R^{1/20} - 1) \cdot 100$ .

## 4 Accounting for Comparative Development

This section extends the baseline model to a multi-country version that allows exploring the implications for patterns of comparative development and cross-country growth studies.

<sup>&</sup>lt;sup>14</sup>Figures A.1 and A.2 in the Online Appendix show corresponding figures for capital returns when normalized by income levels, and when using data from alternative sources and for different subperiods, respectively.

## 4.1 A Multi-Country Version of the Model with Technology Diffusion

To extend the model, we construct a synthetic panel of countries. In particular, we simulate an artificial cross-country panel data set that consists of 114 countries that are identical in terms of all parameters and initial conditions. The only exception is initial skilled productivity  $A_0^S$ , which is assumed to differ for exogenous reasons. A lower  $A_0^S$ implies a lower return to production in the skilled sector and, consequently, a later demographic transition, *ceteris paribus*. To discipline the analysis, we calibrate the synthetic panel to match available data on the onset of the demographic transition in each country (following the classification by Reher, 2004). In the baseline parameterization of the model,  $A_0^S$  was set to generate a demographic transition in 1880. Figure 9 (a) depicts the observed distribution of the onset of the fertility transition in the data.



Figure 9: Distribution of the Onset of the Transition (a) Data: Reher (2004).

To simulate additional countries, we solve (21) for  $A_0^S$  to match the observed distribution of dates of the onset of the demographic transition in the data for these countries. The baseline multi-country simulation assumes that every country follows the same development path as prescribed by the model. In addition, we conduct a simulation for the extended model that accounts for technology diffusion from the world frontier.

Figure 9 (b) depicts the respective distribution of onset dates of the transition in the simulated sample. By construction, the simulated distribution matches the empirical distribution in Figure 9 (a), with minor deviations stemming from the fact that in the simulation the dates are forced to correspond to twenty-year periods.<sup>15</sup>

<sup>&</sup>lt;sup>15</sup>The coarseness of the OLG model with a new generation entering every 20 years implies that each of the simulated countries in the synthetic panel of 114 countries effectively belongs to one of eight groups of countries with transition dates between 1860 and 2000. Also notice that the timing of the onset of the

To account for technology diffusion, we also simulate an extended version of the multicountry model that incorporates an adoption process in the spirit of Nelson and Phelps (1966). Suppose that a country can import the latest technology developed by the technologically most advanced country (corresponding to the world technology frontier in the context of the model),  $\bar{A}_t^U$  and  $\bar{A}_t^S$ , but that the ability to import technology depends on human capital. In particular, consider an extension of the technology dynamics in (37) and (38), where education determines whether the technologies used in the two sectors at time t,  $A_t^U$  and  $\bar{A}_t^S$ , are influenced by the world technology frontier  $\bar{A}_t^U$  and  $\bar{A}_t^S$ ,

$$A_{t+1}^{U} = \left(1 + g_{t+1}^{U}\right) A_{t}^{U} + \rho \left[h\left(e_{t+1}\right) - h_{min}\right] \mathcal{F}\left(A_{t}^{U}, \bar{A}_{t}^{U}, \bar{A}_{t-1}^{U}, \bar{A}_{t-2}^{U}\right)$$
(39)

$$A_{t+1}^{S} = \left(1 + g_{t+1}^{S}\right) A_{t}^{S} + \rho \left[h\left(e_{t}\right) - h_{min}\right] \mathcal{F}\left(A_{t}^{S}, \bar{A}_{t}^{S}, \bar{A}_{t-1}^{S}, \bar{A}_{t-2}^{U}\right)$$
(40)

with  $\rho > 0$  and  $h_{min} = h(0)$ .<sup>16</sup> The function  $\mathcal{F}\left(A_t^i, \bar{A}_t^i, \bar{A}_{t-1}^i, \bar{A}_{t-2}^i\right)$ , given by

$$\mathcal{F} = \left[ (\bar{A}_{t}^{i} - A_{t}^{i}) \cdot \frac{\min\left(\bar{A}_{t-1}^{i} - A_{t}^{i}, 0\right)}{\bar{A}_{t-1}^{i} - A_{t}^{i}} + \max\left(\bar{A}_{t-1}^{i} - A_{t}^{i}, 0\right) \cdot \frac{\min\left(\bar{A}_{t-2}^{i} - A_{t}^{i}, 0\right)}{\bar{A}_{t-2}^{i} - A_{t}^{i}} + \max\left(\bar{A}_{t-2}^{i} - A_{t}^{i}, 0\right) \right]$$

for  $i \in \{U; S\}$  captures the decrease in adoption lags reported by Comin and Mestieri (2018).<sup>17</sup> The parameter  $\rho$  determines the level of convergence. In the data the ratio of average GDP per capita in the G7 between 1980 and 2000 relative to the GDP per capita of the richest country in the G7 in the same time span is around 0.825. In the simulation, we set  $\rho = 0.265$ , which implies the same convergence for the G7 in the simulation.

### 4.2 The Role of the Timing of the Transition

To illustrate the consequences of heterogeneity in the timing of the transition, consider two economies that are identical to the baseline economy in all dimensions. Both economies only differ with respect to one characteristic, namely the initial productivity in the skilled sector  $(A_0^S)$ . In particular, the first of these two economies is the baseline economy with an onset of the demographic transition in 1880. The second economy has a relatively lower productivity in skilled production and correspondingly faces a delayed onset of the transition in 1980.

demographic transition is identical in the baseline version of the multi-country model and in the extended version with technology diffusion.

<sup>&</sup>lt;sup>16</sup>The general representation corresponds to widely used diffusion models, see, e.g., Benhabib and Spiegel (2005) and Stokey (2015).

<sup>&</sup>lt;sup>17</sup>This specification stipulates that economies that are sufficiently close to the world technology frontier (with productivity exceeding that at the frontier a generation earlier) adopt directly from the frontier, whereas economies further from the frontier first have to catch up by adopting technologies that represented the frontier one or two generations earlier. Note that divergence in intensity of use, as in Comin and Mestieri (2018), would be captured by country-specific factors either affecting the level of education or the parameter  $\rho$ .



Figure 10: Simulated Development Path of Two Economies

Figure 10 plots the trajectories for population and output per capita growth for these two economies. In the baseline model, both economies exhibit the same dynamics, although on a different time scale. The delay in the demographic transition entails a delay in income per capita growth. This delay is seen also in the extended model that allows for technology spillovers and in which follower countries are able to import technology from the frontier. This import implies steeper productivity gains the larger the gap to the world frontier at the time of the transition. Consequently, the trajectories of population and output per capita growth differ after the onset of the transition, with the latecomer country exhibiting a sharper growth take-off and higher growth rates.

The shaded area corresponds to the observation window of 1950 – 2010, which covers the usual sample period in empirical growth studies. Observing both economies only during this limited observation period clearly delivers an incomplete and possibly misleading picture of the development path. Moreover, obviously the (implicit or explicit) assumption of a balanced growth path with different growth rates is not warranted during this observation period. Reduced-form estimates are thus likely to be heavily influenced by the observation period for which they are conducted, and the sample on which they are performed. Empirical results obtained with a sample of countries that are mainly observed during the pre-transitional period of their long-run development trajectory are likely to differ from results obtained with a sample of countries that are mainly post-transitional. Notice that this is true even when the maintained assumption about parameter stability across countries is satisfied, since in both the baseline and the extended version of the model the only difference between the two economies is in initial conditions, not structural parameters. We will return to this issue below.

## 4.3 Accounting for the Empirical Patterns of Comparative Development

We are now in a position to assess whether the extended model can account for the empirical patterns of comparative development across countries.

#### 4.3.1 Cross-country Differences in Income and Productivity

The positive association between differences in total factor productivity and income differences across countries constitutes another established empirical fact (Hall and Jones, 1999, Jones and Romer, 2010). In addition, physical capital per worker correlates positively with income per capita while at the same time the returns to physical capital are very similar across countries, despite substantial differences in economic development. This suggests that lower capital ratios in less developed countries are associated with lower levels of complementary factors like human capital (Caselli and Feyrer, 2007). This observation also raised a discussion about the reasons for persistent differences in productivity and the failure of global technology diffusion to foster income convergence. In fact, work by Comin and Hobijn (2010) has shown that a significant part of the cross-country income differences is accounted for by variation in the adoption of technologies. More recently, work by Comin and Mestieri (2018) has documented that a significant part of cross-country income divergence during the 19<sup>th</sup> century can be traced back to delays in the adoption of new technologies, while the cross-country income divergence during the 20<sup>th</sup> century is associated with a divergence in the intensity of use of new technologies. Paralleling the arguments explaining persistent differences in capital intensity, a candidate explanation for this pattern is the lack of complementary factors needed for technology adoption, such as human capital (e.g., Nelson and Phelps, 1966, Zeira, 2009).

Large income and productivity differences across countries are a natural feature of the cross-country version of the model. These differences are primarily the result of differences in the timing of the onset of the demographic transition. As a direct implication of the acceleration of productivity growth due to the expansion of human capital and the associated human capital externalities, the gap in income and productivity across countries initially increases. Hence, the technological frontier displays a sharp acceleration when the forerunner country undergoes the economic and demographic transition.

Figure 11 shows the relationship between comparative development differences in incomes and productivity relative to the US. In the simulation, the US is treated as a country with an education transition in 1920. For the cross-section in the year 2000, the figure demonstrates the model's ability to reflect the well-known positive correlation between income and productivity. However, in the model, income and productivity differences are crucially related to delays in the demographic transition.<sup>18</sup>

 $<sup>^{18}</sup>$  For the reasons discussed in footnote 15, the effective variation when plotting cross-sectional differences



Figure 11: Comparative Differences in Income and TFP (a) Data: Penn World Tables (Feenstra et al., 2015, version 9.1). (b) Model: baseline, artificial cross-country panel, evaluated in 2000.

#### 4.3.2 Cross-country Differences in Capital Intensity and Capital Returns

Figure 12 depicts the comparative development patterns related to capital intensity per worker. The data in Panel (a) show a strong positive association of capital intensity with economic development. This pattern also emerges in the model simulation, as depicted in Panel (b). It is important to notice in this context that the simulated data reflect the relation along the development trajectory. This provides support for the conjecture that the comparative development patterns are consistent with the non-linear dynamics of a unified growth model.



Figure 12: Comparative Development: Capital Intensity per Worker (a) Data: Penn World Tables (Feenstra et al., 2015, version 9.1). (b) Model: baseline, artificial cross-country panel.

Finally, Figure 13 depicts the cross-sectional relationship between income and the marginal product of capital. Panel (a) shows that the marginal product of capital is fairly

across the 114 simulated model economies corresponds to the differences between eight distinct groups of countries that differ with respect to the onset of their demographic transition.

uncorrelated with output per capita, with rich countries even displaying a moderately higher marginal product on average than the poor countries. Panel (b) documents that the model is also able to generate this fact. Despite the larger capital intensity, the return to capital is non-declining as consequence of the complementarity between physical capital and human capital in the model. This is in line with earlier conjectures that poorer countries exhibit lower capital-labor ratios and lower returns to capital as consequence of the low levels of complementary factors such as human capital (see, e.g., Caselli and Feyrer, 2007).



Figure 13: Comparative Differences in the Marginal Product of Capital (a) Data: Caselli and Feyrer (2007, Table II). Returns after correction for natural capital and price differences (*PMPKL*). (b) Model: baseline, artificial cross-country panel, evaluated in 2000, normalized to forerunner country.

### 4.3.3 Growth Variability and Distance to Technology Frontier

The previous discussion is also relevant for a last stylized fact of long-run growth, according to which the variability of growth in incomes per capita increases with the distance to the world technology frontier (Jones and Romer, 2010). According to this observation, economies that have undergone substantial economic development also exhibit lower variation in growth than countries that are still in the early phases of the development process.

The model also captures the observation that the variation in the growth rate of income per capita is higher the greater the distance to the technology frontier. Intuitively, the fact that growth rates exhibit greater variation among the less developed countries is related to the observation that these countries are either in the early phases of their development process and correspondingly experience low but slowly accelerating growth rates, or they are undergoing the demographic and economic transition with the associated acceleration and overshooting of growth rates above the balanced growth path. The extension of the model to technology diffusion implies an additional momentum that contributes to faster growth as consequence of technology diffusion relative to the frontier economy. One way to illustrate this is by relating growth rates to relative incomes as shown in Figure 14. In particular, Panel (a) plots growth rates conditional on relative incomes compared to the frontier for a panel of countries over the period 1950 to 2010 for data from the Penn World Tables. Panel (b) reveals a very similar hump-shaped relation for the corresponding simulated data.<sup>19</sup> The similarities provide support for the conjecture that the frontier country grows faster than pre-transitional countries, whereas countries that are undergoing the transition experience the most rapid growth. These results also complement empirical work that has documented an interaction between education and the distance to the world technology frontier (see, e.g., Madsen, 2014).



Figure 14: Growth Relative to Frontier: Data vs. Simulation

Note: Binscatter plots, lines correspond to a quadratic fit (both panels). Annualized growth rates over twenty-year periods. (a) Data: Penn World Tables (Feenstra et al., 2015, version 9.1, years 1950, 1970, 1990, 2010.). (b) Model: baseline, artificial cross-country panel, evaluated in 1960, 1980, 2000, 2020.

## 5 Implications for Cross-Country Growth Empirics

To further explore the comparative development patterns generated by the model and the resulting implications for the interpretation of empirical observations, the following analysis considers its empirical relevance through the lens of reduced-form regressions. In particular, we investigate whether regressions based on synthetic simulated cross-country panel data deliver similar patterns as regressions based on actual data. In this context it should be noted that the simulated data consist of economies that only differ in terms of an initial condition that determines the timing of their demographic and economic transition, but that are otherwise identical.

<sup>&</sup>lt;sup>19</sup>For the extension to technology diffusion, the qualitative pattern looks similar, see Figure A.3 in the Online Appendix.

## 5.1 Data and Empirical Specification

The empirical analysis is based on cross-country panel data for income, physical capital, and human capital (all in per capita terms) from the Penn World Tables version 9.1 by Feenstra, Inklaar, and Timmer (2015). In these data, physical capital stocks are computed using the perpetual inventory method based on investment data taken from national accounts. Human capital is measured as an index based on average years of schooling from Barro and Lee (2013) and Cohen and Leker (2014). The estimation is conducted for a sample of 114 countries over the period 1950 – 2010, using twenty-year panel data in order to be consistent with the simulation.

We compare the model to the data through the lens of a standard reduced-form empirical growth framework. The reduced-form estimation framework follows the conventional specification used in reduced-form empirical work. In particular, the empirical framework for country i at time t is given by

$$\ln\left(\frac{Y_{i,t}}{N_{i,t}}\right) = a_1 \ln\left(\frac{Y_{i,t-1}}{N_{i,t-1}}\right) + a_2 \ln\left(\frac{K_{i,t}}{L_{i,t-1}}\right) + a_3 h_{i,t} + \mu_i + b_t + u_{i,t}$$

where  $\frac{Y_{i,t}}{N_{i,t}}$  corresponds to income per capita,  $\frac{K_{i,t}}{L_{i,t-1}}$  is the capital endowment per worker, and  $h_{i,t}$  is an index measure of average individual human capital. To account for timeinvariant differences across countries that might influence the level of development, such as the initial productivity in the skilled sector, the estimation includes a full set of country fixed effects  $\mu_i$ . Additionally, to capture time-variant, global dynamics such as the world technology frontier the estimation also includes a full set of common time dummies  $b_t$ . The variables  $a_1$ ,  $a_2$ , and  $a_3$  are coefficients to be estimated.

To explore whether the estimation reveals the mechanics behind the development dynamics, in particular the crucial role of the demographic transition for the relevance of human capital, we also estimate an extended specification of the empirical model that accounts for heterogeneity in the coefficients of interest before and after the onset of the demographic transition. In the data, we use the same classification of transition dates by Reher (2004) as in the construction of the synthetic panel discussed above.

### 5.2 Accounting for Cross-Country Growth Empirics

Table 2 presents the estimates. Columns (1) and (2) show the respective estimation results for the empirical data organized as twenty-year panel and for a specification with only physical capital, human capital, a lagged dependent variable and two-way fixed effects, thus exploiting variation within countries. Columns (3) and (4) present the corresponding results for the synthetic panel of simulated countries using the baseline specification of the model without technology diffusion. Columns (5) and (6) show the corresponding results for the extended model with technology diffusion.

	Dependent variable: ln [Income p/c]					
	Data 20-year panel		Baselin	e Model	Extension	
	(1) (2)		(3)	(4)	(5)	(6)
GDP p.c. (t-1)	$0.090 \\ (0.07)$	$0.059 \\ (0.07)$	$-0.059^{*}$ (0.03)	$\begin{array}{c} 0.082^{***} \\ (0.00) \end{array}$	-0.024 (0.02)	$\begin{array}{c} 0.14^{***} \\ (0.04) \end{array}$
k	$0.47^{***}$ (0.07)		$0.79^{***}$ (0.04)		$0.72^{***}$ (0.02)	
h	$0.54^{**}$ (0.25)		$0.22^{***}$ (0.03)		$0.46^{***}$ (0.02)	
k (pre-transition)		$\begin{array}{c} 0.48^{***} \\ (0.07) \end{array}$		$0.67^{***}$ (0.00)		$0.59^{***}$ (0.04)
k (post-transition)		$0.46^{***}$ (0.07)		$0.60^{***}$ (0.00)		$0.50^{***}$ (0.05)
h (pre-transition)		$0.49 \\ (0.73)$		0 (.)		0(.)
h (post-transition)		$0.60^{**}$ (0.25)		$\begin{array}{c} 0.36^{***} \ (0.00) \end{array}$		$0.62^{***}$ (0.04)
Year and Country FE	Yes	Yes	Yes	Yes	Yes	Yes
Observations $R^2$ (within)	$269 \\ 0.78$	$269 \\ 0.79$	342 1.00	342 1.00	$342 \\ 1.00$	$342 \\ 1.00$

Table 2: Reduced-Form Estimates Reconsidered: Data vs. Simulated Data

Results of fixed effects regressions. Columns (1) and (2): The dependent variable is (the log of) real GDP at constant 2011 national prices (in mil. 2011 US \$) divided by population size (in millions), k is (the log of) capital stock at constant 2011 national prices (in mil. 2011 US \$) divided by the number of persons employed (in millions), h is a human capital index, normalized to lie between 0 and 1; for details see data description of the PWT 9.1. Sample period: twenty-year panel {1950, 1970, ..., 2010}. Columns (3)-(6): The dependent variable is (the log of) income per capita, k is (the log of) capital endowment per worker, and h is constructed as an index from the (logged) human capital measure normalized to lie between 0 and 1, replicating the index in the data. Sample period: twenty-year panel {1960, 1980, ..., 2020}. All specifications include a full set of country fixed effects and common time dummies. Standard errors are clustered at the country level. \*, \*\*, \*\*\* denote significance at 10%, 5%, and 1% levels.

Several aspects are noteworthy. First, the data estimates reveal coefficients for physical capital and human capital that roughly correspond to the usual estimates in terms of an elasticity for physical capital and human capital of 0.47 and 0.5 - 0.6, depending on the specification. When allowing for heterogeneous effects of physical capital and human capital before and after the onset of the demographic transition in column (2), the data estimates reveal little evidence for any asymmetry in physical capital. For human capital, on the other hand, the results only document a significant positive effect post-demographic

transition.

The corresponding estimates for the simulated data reveal a striking similarity to the estimates obtained with actual data. In particular, regardless of whether one considers the baseline model or the extension, the estimates obtained from the synthetic panel are qualitatively and quantitatively comparable. The effect of physical capital reveals no heterogeneity before and after the onset of the demographic transition, whereas, by construction, human capital only reveals variation that allows for identifying an effect after the transition. The effect of human capital in the most parsimonious specification is slightly smaller in size, but for more extended specifications that account for non-linear dynamics, the overall pattern becomes very comparable. In addition to the similarity in coefficient estimates, the results also reveal a similarity in the relative contributions of the different variables in terms of explanatory power.<sup>20</sup>

Taken together, this evidence suggests that despite the non-linear influence of the demographic transition, the simulated data reveal estimation results that resemble the familiar patterns from analyses in growth empirics. The results are also consistent with a mechanism that works through an acceleration in the accumulation of human capital, which is influenced by the demographic transition as well as by the associated dynamics in the productivity environment, thereby reflecting insights that have been expressed by Hsieh and Klenow (2010). It is worth emphasizing, however, that the striking similarity has been obtained with a synthetic panel of simulated data in which cross-country variation only emerges through the (country-specific and time-invariant) initial condition of skilled productivity,  $A_0^S$ . The results show that an empirical specification with two-way fixed effects does not account appropriately for development differences that manifest themselves in delays of non-linear transition dynamics. However, even though the reduced-form empirical estimation model does not explicitly account for the non-linear dynamics featured by the theoretical model, the estimates are very similar. Moreover, the synthetic data have not been calibrated in any way to match the estimates. Instead, they were obtained exclusively by matching the transition dates with an otherwise identical parameterization.

In sum, these results can be understood as a proof of concept regarding the suitability of a unified growth model as a data generating process, which, despite the highly non-linear development dynamics, delivers reduced-form estimates that resemble those obtained with empirical data. The evidence is consistent with the conjecture of the demographic transition as a significant turning point in a country's economic development, and with the hypothesis of an acceleration of growth within countries after the onset of the transition. The results demonstrate that the non-linearity of the long-run development trajectory need not necessarily lead to unstable coefficient estimates, which is reassuring for the existing empirical growth literature that is based, at least implicitly, on

 $<sup>^{20}</sup>$ See Table A.1 in the Online Appendix for details.

the notion of a balanced growth path. At the same time, however, the model indicates that comparative development differences can be traced to differences in the timing of the transition, which are themselves the consequence of differences in initial conditions. Moreover, even accounting for technology diffusion across countries does not affect this conclusion. The results therefore illustrate the empirical compatibility of unified growth theories with existing work on growth accounting, while documenting the important role of the timing of the demographic transition for long-run development.

### 5.3 Implications for Technology Diffusion

The results of the simulation of the extended model with cross-country technology diffusion also have implications for the interpretation of unified growth frameworks in the context of contemporaneous comparative development patterns. As illustrated in the last section, the incorporation of technology diffusion across countries does not affect the empirical relevance or the qualitative fit of the model in the context of a synthetic panel analysis. In fact, even the quantitative fit of the model is not adversely affected by the incorporation of diffusion in terms of a positive  $\rho$ .<sup>21</sup>

The discussion of the central role of the demographic transition for the patterns of comparative development raises additional implications in the context of technology diffusion. In particular, the formulation of the diffusion process suggests that the education expansion might play an important role for the adoption and diffusion of technologies that has not been fully appreciated. Considering the diffusion process as specified in (39) and (40) suggests that the level of education, and hence the timing of the demographic transition, might be an important factor in explaining adoption lags and diverging intensity of technology use.

To illustrate this implication, we replicate the analysis of Comin and Mestieri (2018) and extend their specification to allow for heterogeneity in the timing of the demographic transition to affect delays in technology adoption and the intensity of use. The data contains information about the diffusion of 25 major technologies across countries in a variety of sectors (transportation, communication and IT, industrial, agricultural, and medical sectors). The diffusion of these technologies covers a broad set of countries and the invention dates of these technologies are fairly evenly distributed over the past 200 years, with more than half of the technologies in the data being invented during the 19<sup>th</sup> century. The analysis focuses on two elements: the lag with which a particular technology is adopted by a country after the technology has first been invented and the intensity with which the technology is used.<sup>22</sup>

The results for technology-country pairs as unit of observation are shown in Table 3.

<sup>&</sup>lt;sup>21</sup>For illustration purposes, we presented a version with no diffusion that corresponds to a simulation of the baseline model ( $\rho = 0$ ) and an extension of the model with diffusion ( $\rho = 0.265$ ), respectively.

 $<sup>^{22}</sup>$ Both variables are in logs, see Comin and Mestieri (2018) for details on the construction and coding.

Columns (1) and (4) replicate the main results of Comin and Mestrieri (2018, Table 3) and show that adoption lags decline for later invention dates, while also the intensity of use declined, respectively. In line with the hypothesis, the results in columns (2) and (3) also document that adoption lags are larger for countries that exhibit a later demographic transition.<sup>23</sup> Similarly, the results in columns (5) and (6) show that the intensity of use is systematically lower in countries that exhibit a later onset of the demographic transition. Hence, countries with later transitions exhibit a significantly greater adoption lag and a significantly lower intensity of use, above and beyond the trend relative to the invention dates. In other words, the adoption occurs quicker and the intensity of use of adopted technologies is higher in countries that experienced an earlier demographic transition relative to the time of the appearance of new technologies.<sup>24</sup> Moreover, accounting for the demographic transition implies a substantial increase in the explanatory power of the model (in terms of  $R^2$ ), particularly when considering the intensity of use. In other words, the timing of the demographic transition is an important, yet so far largely neglected, factor in the process of technology diffusion.

Replicating the same estimation on the basis of the synthetic panel data obtained with the model extension to technology diffusion delivers qualitatively identical and quantitatively fairly similar results.<sup>25</sup> This is notable since none of the parameters of the model was calibrated on this dimension of the model.

## 6 Conclusion

In this article, we developed a comprehensive unified growth framework that is able to replicate quantitatively the stylized facts of long-run growth dynamics and of comparative development. The research contributes to the literature by developing an expanded unified growth model that incorporates endogenous savings and a capital-skill complementarity. This model not only performs well in terms of matching various historical facts of longrun growth dynamics, including ones that have been documented recently, but can also be extended to a multi-country setting by incorporating technology diffusion from the world technology frontier in order to lend insights for comparative development and crosscountry growth empirics. The expanded unified growth model accounts for the longrun trajectories of income, physical capital, population, and human capital and their association with the transition from stagnation to growth. The model thereby sheds light on the mechanisms that link the empirical facts of long-run growth and comparative

<sup>&</sup>lt;sup>23</sup>In fact, unreported results reveal that the adoption lags are significantly smaller for countries that had already undergone the demographic transition at the time of invention, while this effect is weaker the later the invention date, as reflected by a positive interaction term.

<sup>&</sup>lt;sup>24</sup>Table A.2 in the Online Appendix shows that the main result also holds when splitting the sample into Western and Non-Western countries, as in the analysis of Comin and Mestieri (2018).

 $<sup>^{25}\</sup>mbox{See}$  Table A.3 in the Online Appendix.

	Western Countries:							
Dependent Variable:	Log	g [Adoption	Lag]	Log [Intensity]				
	(1)	(2)	(3)	(4)	(5)	(6)		
Invention Year-1820	$-0.011^{***}$ (0.00)	$-0.0097^{***}$ (0.00)	$-0.0096^{***}$ (0.00)	$-0.0016^{**}$ (0.00)	$-0.0039^{***}$ (0.00)	$-0.0038^{***}$ (0.00)		
Year of Demogr. Tr.		$\begin{array}{c} 0.0074^{***} \\ (0.00) \end{array}$			$-0.015^{***}$ (0.00)			
Follower Country			$0.51^{***}$ (0.11)			$-0.35^{*}$ (0.20)		
Trailer Country			$0.32^{***}$ (0.08)			$-0.94^{***}$ (0.13)		
Latecomer Country			$0.68^{***}$ (0.08)			$-1.32^{***}$ (0.19)		
$\begin{array}{c} \text{Observations} \\ R^2 \end{array}$	$1009 \\ 0.41$	$\frac{1009}{0.47}$	$\begin{array}{c} 1009 \\ 0.48 \end{array}$	$\frac{1039}{0.01}$	$\begin{array}{c} 1039\\ 0.36\end{array}$	$\frac{1039}{0.38}$		

 Table 3: Technology Diffusion

Columns (1) and (4) replicate the results in Table 3, columns (1) and (4) of Comin and Mestrieri (2018). Country labels are as in Reher (2004): Forerunner countries (reference category) experienced the fertility transition before 1935, Followers between 1935 and 1964, Trailers between 1965 and 1979, and Latecomers after 1980. Robust standard errors clustered at the country level are in parentheses. Each observation is re-weighted so that each technology carries equal weight. \*, \*\*, \*\*\* denote significance at 10%, 5%, and 1% levels.

development to each other.<sup>26</sup>

On the quantitative side, the paper documents that a parsimoniously parameterized quantitative version of the model that is calibrated to population and income dynamics in England during the 18<sup>th</sup> and 19<sup>th</sup> century is able to account for a large set of disparate empirical regularities. An extension to a multi-country setting in which countries only differ in the timing of the demographic transition is able to account for patterns of comparative development. Analyzing the model performance through the lens of the standard reduced-form empirical growth approach documents that, despite the highly non-linear dynamics and the out-of-sample nature of this exercise, the regression results closely resemble those obtained with standard cross-country panel data. Moreover, model simulations show that the consideration of international technology diffusion does not affect the qualitative results regarding the model dynamics. However, the extension has implications for explaining the recently reported regularities regarding the lags and intensity with which new technologies are adopted.

<sup>&</sup>lt;sup>26</sup>In this sense, the paper accomplishes a long-standing quest in the growth literature, delivering a comprehensive growth framework that fits the stylized facts and that captures "the endogenous accumulation of, and interaction between, (...) four state variables: ideas, population, and human capital", in addition to physical capital, as requested by Jones and Romer (2010) at the end of their survey of the facts of long-run growth (Jones and Romer, 2010, p. 226).

The results of this article provide support for the hypothesis that the unified growth approach is able to rationalize the stylized facts of long-run growth. The findings suggest that the timing of the demographic transition is a critical watershed for long-run growth dynamics that affects patterns contained in cross-country panel data. Consistent with the implications from the unified growth literature, the results also suggest that countries that experience an earlier take-off from stagnation to growth are richer and develop faster. Nevertheless, the momentum of the transition eventually fades out, which provides an explanation for the observed slow-down in growth among advanced economies.

The findings suggest that the usefulness of the insights of unified growth theory for the empirical growth literature have not been adequately appreciated. The intention of this paper was to go beyond the standard methodology of illustrative simulations of the non-linear long-run dynamics of a unified growth model, and use the model instead as a data generating process to study the facts of long-run growth. More work is needed to explore the implications of heterogeneity in the timing of the demographic transition as a determinant of comparative development patterns. In light of the results shown in this paper, this aspect has not received sufficient attention in the existing literature. The results for the extension to technology diffusion indicate the relevance of unified growth mechanisms for understanding the patterns of technology adoption, but more work is needed to assess the quantitative importance of accounting for cross-country spillovers. A more structural approach to other types of spillovers, e.g., related to capital mobility, labor mobility, or trade in intermediate or final goods constitutes an interesting direction for future work.

## References

- ALLEN, R. C. (2009): "Engels' pause: Technical change, capital accumulation, and inequality in the british industrial revolution," *Explorations in Economic History*, 46(4), 418–435.
- ALVAREZ-NOGAL, C., AND L. PRADOS DE LA ESCOSURA (2013): "The rise and fall of Spain (1270–1850)," *Economic History Review*, 66(1), 1–37.
- ASHRAF, Q., AND O. GALOR (2011): "Dynamics and stagnation in the Malthusian epoch," *American Economic Review*, 101(5), 2003–41.
- BAR, M., AND O. LEUKHINA (2010): "Demographic Transition and Industrial Revolution: A Macroeconomic Investigation," *Review of Economic Dynamics*, 13(2), 424–451.
- BARRO, R., AND J.-W. LEE (2013): "A New Data Set of Educational Attainment in the World, 1950-2010," *Journal of Development Economics*, 104, 184–198.
- BAUDIN, T., AND R. STELTER (2022): "The rural exodus and the rise of Europe," *Journal of Economic Growth*, 27(3), 365–414.
- BECKER, G. S., K. M. MURPHY, AND R. TAMURA (1990): "Human capital, fertility, and economic growth," *Journal of Political Economy*, 98(5, Part 2), S12–S37.
- BECKER, S. O., E. HORNUNG, AND L. WOESSMANN (2011): "Education and catchup in the industrial revolution," *American Economic Journal: Macroeconomics*, 3(3), 92–126.
- BENHABIB, J., AND M. SPIEGEL (2005): "Human Capital and Technology Diffusion," in Handbook of Economic Growth, ed. by P. Aghion, and S. Durlauf, chap. 13, pp. 936–966. Elsevier Science B.V., Amsterdam.
- BETRAN, C., J. F., AND M. A. PONS (2010): "Explaining UK wage inequality in the past globalization period, 1880–1913," *Cliometrica*, 4(1), 19–50.
- BILLARI, F. C., AND H.-P. KOHLER (2004): "Patterns of Low and Lowest-Low Fertility in Europe," *Population Studies*, 58(2), 161–176.
- BOLT, J., R. INKLAAR, H. DE JONG, AND J. L. VAN ZANDEN (2018): "Maddison project database, version 2018," *Rebasing 'Maddison': new income comparisons and the shape of long-run economic development.*
- BOLT, J., AND J. L. VAN ZANDEN (2020): "Maddison style estimates of the evolution of the world economy. A new 2020 update," *Maddison-Project Working Paper WP-15*, *University of Groningen, Groningen, The Netherlands.*

- BONGAARTS, J. (2002): "The End of the Fertility Transition in the Developed World," Population and Development Review, 28(3), 419–443.
- BROADBERRY, S., B. M. CAMPBELL, A. KLEIN, M. OVERTON, AND B. VAN LEEUWEN (2015): British economic growth, 1270–1870. Cambridge University Press.
- BROADBERRY, S. N., AND A. DE PLEIJT (2021): "Capital and Economic Growth in Britain 1270–1870," *CEPR Discussion Paper*, 15889.
- CASELLI, F., AND J. FEYRER (2007): "The Marginal Product of Capital," *Quarterly Journal of Economics*, 122(2), 535–568.
- CERVELLATI, M., G. MEYERHEIM, AND U. SUNDE (2019): "The timing of the demographic transition and economic growth," *Economics Letters*, 181, 43–46.
- CERVELLATI, M., AND U. SUNDE (2005): "Human Capital, Life Expectancy, and the Process of Development," *American Economic Review*, 95(5), 1653–1672.
- (2011a): "Disease and Development: The Role of Life Expectancy Reconsidered," *Economics Letters*, 113, 269–272.
- (2011b): "Life Expectancy and Economic Growth: The Role of the Demographic Transition," *Journal of Economic Growth*, 16, 99–133.
- (2015): "The Economic and Demographic Transition, Mortality, and Comparative Development," *American Economic Journal: Macroeconomics*, 7(3), 1–39.
- CERVELLATI, M., U. SUNDE, AND K. F. ZIMMERMANN (2017): "Demographic dynamics and long-run development: insights for the secular stagnation debate," *Journal of Population Economics*, 30, 401–432.
- CLARK, G. (2010): "The Macroeconomic Aggregates for England, 1209–2008," *Research in Economic History*, 27, 51–140.
- COHEN, D., AND L. LEKER (2014): "Health and Education: Another Look with the Proper Data," *mimeo, Paris School of Economics.*
- COMIN, D., AND B. HOBIJN (2010): "An Exploration of Technology Diffusion," *American Economic Review*, 100(5), 2031–2059.
- COMIN, D., AND M. MESTIERI (2018): "If Technology Has Arrived Everywhere, Why Has Income Diverged?," *American Economic Journal: Macroeconomics*, 10(3), 137– 178.
- CRAFTS, N. (2021): "Understanding productivity growth in the industrial revolution," *Economic History Review*, 74(2), 309–338.

- CRAFTS, N., AND T. C. MILLS (2022): "Considering the Counterfactual: Real Wages in the First Industrial Revolution," *Economic Journal*, 132(645), 1994–2006.
- DALGAARD, C.-J., AND H. STRULIK (2013): "The History Augmented Solow model," European Economic Review, 63(1), 134–149.
- DELVENTHAL, M. J., J. FERNÁNDEZ-VILLAVERDE, AND N. GUNER (2021): "Demographic Transitions Across Time and Space," *NBER Working Paper*, 29480.
- DOEPKE, M. (2004): "Accounting for Fertility Decline During the Transition to Growth," Journal of Economic Growth, 9(3), 347–383.
- FEENSTRA, R. C., R. INKLAAR, AND M. P. TIMMER (2015): "The Next Generation of the Penn World Table," *American economic review*, 105(10), 3150–82.
- FEINSTEIN, C. H. (1998): "Pessimism perpetuated: real wages and the standard of living in Britain during and after the industrial revolution," *Journal of Economic History*, 58(3), 625–658.
- FOUQUET, R., AND S. BROADBERRY (2015): "Seven centuries of European economic growth and decline," *Journal of Economic Perspectives*, 29(4), 227–44.

GALOR, O. (2011): Unified Growth Theory. Princeton University Press, Princeton, N.J.

(2012): "The demographic transition: causes and consequences," Cliometrica, 6(1), 1–28.

- (2022): The Journey of Humanity: The Origins of Wealth and Inequality. Penguin Random House, New York, N.Y.
- GALOR, O., AND O. MOAV (2006): "Das Human Kapital: A Theory of the Demise of the Class Structure," *Review of Economic Studies*, 73(1), 85–117.
- GALOR, O., O. MOAV, AND D. VOLLRATH (2009): "Inequality in Land Ownership, the Emergence of Human Capital Promoting Institutions, and the Great Divergence," *Review of Economic Studies*, 76(1), 143–179.
- GALOR, O., AND A. MOUNTFORD (2008): "Trading Population for Productivity: Theory and Evidence," *Review of Economic Studies*, 75(4), 1143–1179.
- GALOR, O., AND D. N. WEIL (2000): "Population, Technology, and Growth: From Malthusian Stagnation to the Demographic Transition and Beyond," American Economic Review, 90(4), 807–828.

- GENNAIOLI, N., R. LA PORTA, F. LOPEZ-DE SILANES, AND A. SHLEIFER (2012): "Human Capital and Regional Development," *Quarterly Journal of Economics*, 128(1), 105–164.
- GOLDBERG, P. K., AND N. PAVCNIK (2007): "Distributional Effects of Globalization in Developing Countries," *Journal of Economic Literature*, 45(1), 39–82.
- GOLDIN, C., AND L. F. KATZ (2007a): *The Race Between Education and Technology*. Harvard University Press, Cambridge, Mass.
- (2007b): "The Race Between Education and Technology: The Evolution of U.S. Educational Wage Differentials 1890 to 2005," *NBER Working Paper*, 12984.
- GOLDSTEIN, J. R., T. SOBOTKA, AND A. JASILIONIENE (2009): "The End of "Lowest-Low" Fertility?," *Population and Development Review*, 35(4), 663–699.
- GORDON, R. J. (2012): "Is U.S. Economic Growth Over? Faltering Innovations Confront the Six Headwinds," *NBER Working Paper*, 18315.
- (2014): "The Demise of U.S. Economic Growth: Restatement, Rebuttal, and Reflections," *NBER Working Paper*, 18315.
- (2016): The Rise and Fall of American Growth. Princeton University Press, Princeton, N.J.
- GREENWOOD, J., N. GUNER, AND R. MARTO (2021): "The Great Transition: Kuznets Facts for Family-Economists," University of Pennsylvania Population Center Working Paper (PSC/PARC), 2021-65.
- GREENWOOD, J., AND A. SEHSHADRI (2002): "The U.S. Demographic Transition," American Economic Review, 92(2), 153–159.
- GUNDLACH, E., AND M. PALDAM (2020): "A hump-shaped transitional growth path as a general pattern of long-run development," *Economic Systems*, 44(3), 100809.
- HALL, R. E., AND C. I. JONES (1999): "Why Do Some Countries Produce So Much More Output Per Worker Than Others?," *Quarterly Journal of Economics*, 114(1), 83–116.
- HANSEN, G. D., AND E. C. PRESCOTT (2002): "Malthus to Solow," American Economic Review, 92(4), 1205–1217.
- HANUSHEK, E. A., J. RUHOSE, AND L. WOESSMANN (2017): "Knowledge Capital and Aggregate Income Differences: Development Accounting for US States," *American Economic Journal: Macroeconomics*, 9(4), 184–224.

- HSIEH, C.-T., AND P. J. KLENOW (2010): "Development Accounting," American Economic Journal: Macroeconomics, 2(1), 207–223.
- HUMPHRIES, J., AND J. WEISDORF (2019): "Unreal wages? Real income and economic growth in England, 1260–1850," *Economic Journal*, 129(623), 2867–2887.
- JOHNSON, P., AND C. PAPAGEORGIOU (2020): "What remains of cross-country convergence?," Journal of Economic Literature, 58(1), 129–75.
- JONES, C. I., AND P. ROMER (2010): "The New Kaldor Facts: Ideas, Institutions, Population, and Human Capital," *American Economic Journal: Macroeconomics*, 2(1), 224–245.
- JORDÀ, O., K. KNOLL, D. KUSHNIKOV, M. SCHULARICK, AND A. M. TAYLOR (2019): "The Rate of Return on Everything, 1870–2015," *Quarterly Journal of Economics*, 134(3), 1225–1298.
- KALDOR, N. (1957): "A model of economic growth," *Economic Journal*, 67(268), 591–624.
- KALEMLI-OZCAN, S. (2003): "A Stochastic Model of Mortality, Fertility, and Human Capital Investment," *Journal of Development Economics*, 70(1), 103–118.
- KREMER, M. (1993): "Population Growth and Technological Change: One Million B.C. to 1990," *Quarterly Journal of Economics*, 108(3), 681–716.
- KRUSELL, P., L. E. OHANIAN, J.-V. RIOS-RULL, AND G. L. VIOLANTE (2000): "Capital-Skill Complementarity and Inequality: A Macroeconomic Analysis," *Econometrica*, 68(5), 1029–1053.
- LEE, J.-W., AND H. LEE (2016): "Human capital in the long run," Journal of Development Economics, 122, 147–169.
- LINDNER, I., AND H. STRULIK (2020): "Innovation and inequality in a small world," International Economic Review, 61(2), 683–719.
- MADSEN, J. B. (2014): "Human Capital and the World Technology Frontier," *Review of Economics and Statistics*, 96(4), 676–692.
- MALANIMA, P. (2011): "The long decline of a leading economy: GDP in central and northern Italy, 1300–1913," *European Review of Economic History*, 15(2), 169–219.
- MANKIW, N., D. ROMER, AND D. WEIL (1992): "A Contribution to the Empirics of Economic Growth," *Quarterly Journal of Economics*, 107(2), 407–437.

- MYRSKYLÄ, M., H.-P. KOHLER, AND F. C. BILLARI (2009): "Advances in development reverse fertility declines," *Nature*, 460(7256), 71–743.
- NELSON, R., AND E. PHELPS (1966): "Investment in Humans, Technological Diffusion, and Economic Growth," *American Economic Review*, 51(2), 69–75.
- OECD (2015): Escaping the Stagnation Trap: Policy Options for the Euro Area and Japan. Organization for Economic Cooperation and Development, Paris.
- PALMA, N., AND J. REIS (2019): "From convergence to divergence: Portuguese economic growth, 1527–1850," The Journal of Economic History, 79(2), 477–506.
- PRESTON, S. H., P. HEUVELINE, AND M. GUILLOT (2001): Demography Measuring and Modeling Population Processes. Blackwell, Oxford.
- RANGAZAS, P. (2002): "The Quantity and Quality of Schooling and U.S. Productivity Growth (1870–2000)," *Review of Economic Dynamics*, 5, 932–964.
- REHER, D. S. (2004): "The Demographic Transition Revisited as a Global Process," *Population, Space and Place*, 10(1), 19–41.
- SCHMELZING, P. (2020): "Eight centuries of global real interest rates, R-G, and the 'suprasecular' decline, 1311–2018," *Bank of England Staff Working Paper*, 845.
- SCHÖN, L., AND O. KRANTZ (2015): "New Swedish Historical National Accounts since the 16th Century in Constant and Current Prices," Discussion Paper 140, Lund University, Department of Economic History.
- STOKEY, N. L. (2015): "Catching Up and Falling Behind," *Journal of Economic Growth*, 20(1), 1–36.
- STOKEY, N. L. (2021): "Technology and skill: Twin engines of growth," *Review of Economic Dynamics*, 40, 12–43.
- STRULIK, H., K. PRETTNER, AND A. PRSKAWETZ (2013): "The past and future of knowledge-based growth," *Journal of Economic Growth*, 18(4), 411–437.
- STRULIK, H., AND J. WEISDORF (2008): "Population, Food, and Knowledge: A Simple Unified Growth Theory," *Journal of Economic Growth*, 13(3), 195–216.
- SUNDE, U., T. DOHMEN, B. ENKE, A. FALK, D. HUFFMAN, AND G. MEYERHEIM (2022): "Patience and comparative development," *Review of Economic Studies*, 89(5), 2806–2840.
- SUNDE, U., AND T. VISCHER (2015): "Human Capital and Growth: Specification Matters," *Economica*, 82(326), 368–390.

- TEULINGS, C., AND R. BALDWIN (eds.) (2014): Secular Stagnation: Facts, Causes, and Cures. CEPR Press, London.
- TURVEY, R. (2010): "The cost of living in London, 1740-1834," *LSE Working Paper*, 147/10.
- VAN ZANDEN, J. L. (2009): "The skill premium and the 'Great Divergence'," *European Review of Economic History*, 13(1), 121–153.
- VAN ZANDEN, J. L., AND B. VAN LEEUWEN (2012): "Persistent but not consistent: The growth of national income in Holland 1347–1807," *Explorations in economic history*, 49(2), 119–130.
- ZEIRA, J. (2009): "Why and how education affects economic growth," *Review of International Economics*, 17(3), 602–614.

## A Appendix

## A.1 Derivation of Generic Growth Rates

Unskilled Sector Growth in the unskilled sector is given by

$$1 + g_{t+1}^{Y_U} = \frac{Y_{t+1}^U}{Y_t^U}$$

Updating (2) by one period and using (13) gives

$$1 + g_{t+1}^{Y_U} = \frac{L_{t+1}^{\alpha} \left( A_{t+1}^U X \right)^{1-\alpha}}{L_t^{\alpha} \left( A_t^U X \right)^{1-\alpha}} = \frac{\left( n_t L_t \right)^{\alpha} \left( A_{t+1}^U X \right)^{1-\alpha}}{L_t^{\alpha} \left( A_t^U X \right)^{1-\alpha}} = n_t^{\alpha} (1 + g_{t+1}^{A_U})^{1-\alpha}$$

Skilled Sector Growth in the skilled sector is given by

$$1 + g_{t+1}^{Y_S} = \frac{Y_{t+1}^S}{Y_t^S}$$

Updating (3) by one period and using (13), (14), and (15) gives

$$1 + g_{t+1}^{Y_S} = \frac{H_{t+1}^{\beta} \left(A_{t+1}^S K_{t+1}\right)^{1-\beta}}{H_t^{\beta} \left(A_t^S K_t\right)^{1-\beta}} = \frac{\left(h\left(e_{t+1}\right) L_t\right)^{\beta} \left(A_{t+1}^S s_t L_t\right)^{1-\beta}}{\left(h\left(e_t\right) L_{t-1}\right)^{\beta} \left(A_t^S s_{t-1} L_{t-1}\right)^{1-\beta}}$$
$$= \frac{n_{t-1} L_{t-1}}{L_{t-1}} \left(\frac{h\left(e_{t+1}\right)}{h\left(e_t\right)}\right)^{\beta} \left(\frac{A_{t+1}^S}{A_t^S}\right)^{1-\beta} \left(\frac{s_t}{s_{t-1}}\right)^{1-\beta} = n_{t-1} (1 + g_{t+1}^h)^{\beta} (1 + g_{t+1}^{A_S})^{1-\beta} (1 + g_t^s)^{1-\beta}$$

**Population** Population growth is given by

$$1 + g_{t+1}^N = \frac{N_{t+1}}{N_t}$$

Substituting (16) and using (13) gives

$$1 + g_{t+1}^N = \frac{L_t + L_{t+1}}{L_{t-1} + L_t} = \frac{(1+n_t)L_t}{(1+n_{t-1})L_{t-1}} = \frac{(1+n_t)n_{t-1}L_{t-1}}{(1+n_{t-1})L_{t-1}} = \frac{n_{t-1}(1+n_t)}{1+n_{t-1}}$$

Output per Capita Output per capita growth is given by

$$1 + g_{t+1}^y = \frac{y_{t+1}}{y_t} \equiv \frac{\frac{Y_{t+1}}{N_{t+1}}}{\frac{Y_t}{N_t}}$$

Re-arranging and using (1) gives

$$1 + g_{t+1}^y = \frac{Y_{t+1}}{Y_t} \frac{N_t}{N_{t+1}} = \frac{Y_{t+1}^U + Y_{t+1}^S}{Y_t} \frac{N_t}{N_{t+1}} = \left(\frac{Y_{t+1}^U}{Y_t^U} \frac{Y_t^U}{Y_t} + \frac{Y_{t+1}^S}{Y_t^S} \frac{Y_t^S}{Y_t}\right) \frac{N_t}{N_{t+1}}$$

Using 
$$\frac{Y_{t+1}^U}{Y_t^U} = 1 + g_{t+1}^{Y_U}, \frac{Y_{t+1}^S}{Y_t^S} = 1 + g_{t+1}^{Y_S}, \theta_t \equiv \frac{Y_t^S}{Y_t}, \text{ and } \frac{N_{t+1}}{N_t} = 1 + g_{t+1}^N \text{ yields}$$
  
$$1 + g_{t+1}^y = \frac{(1 + g_{t+1}^{Y_U})(1 - \theta_t) + (1 + g_{t+1}^{Y_S})\theta_t}{1 + g_{t+1}^N}$$

## A.2 Growth During the Phases of Development

**Malthusian Phase** Notice that since  $s_{t+1} > 0$  and  $n_{t+1} > 0$  it must hold generically that

$$\frac{s_{t+1}}{s_t} + \frac{n_{t+1}}{n_t} > \frac{s_{t+1} + n_{t+1}}{s_t + n_t} = \frac{w_{t+1}^L}{w_t^L}$$

where the last equality follows from (7).

To show the claim that savings dynamics depend on the relative sizes of fertility and productivity growth, consider the limit case in which  $s_t \to w_t^L$  so that all productivity gains are absorbed by savings and not fertility. Combining (4) and (13), we have

$$\frac{s_{t+1}}{s_t} = \left(\frac{1+g_{t+1}^{A_U}}{n_t}\right)^{1-\alpha}$$

To derive the bound for growth in income per capita, note that combining this condition with  $\frac{n_{t+1}}{n_t} = 1$  and using (26), (27), as well as (28) delivers the level of fertility consistent with a Malthusian steady state, denoted by  $\bar{n}_M$ , as

$$\bar{n}_M = \frac{1 + g_{t+1}^{A_U}}{(1 + g_{t+2}^{A_S})^{\frac{1-\beta}{\beta(1-\alpha)}}} = \frac{\phi_U}{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}}$$

In this Malthusian steady state, productivity growth, wage growth, and savings growth exactly balance each other, implying constant fertility.

The per capita growth rate of the unskilled sector is given by

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} = \frac{n_t^{\alpha}(1+g_{t+1}^{A_U})^{1-\alpha}}{1+g_{t+1}^N}$$

Inserting (24) yields

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} = n_t^{\alpha} (1+g_{t+1}^{A_U})^{1-\alpha} \frac{1+n_{t-1}}{n_{t-1}(1+n_t)}$$

In the Malthusian steady state  $n_{t-1} = n_t = \bar{n}_M = \frac{\phi_U}{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}}$  and  $1 + g_{t+1}^{A_U} = \phi_U$ , therefore

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} = \left(\frac{1+g_{t+1}^{A_U}}{\bar{n}_M}\right)^{1-\alpha} = \left(\frac{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}}{\phi_U}\phi_U\right)^{1-\alpha} = \phi_S^{\frac{1-\beta}{\beta}}$$

Conversely, the per capita growth rate of the skilled sector is given by

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = \frac{n_{t-1}(1+g_{t+1}^h)^\beta (1+g_{t+1}^{A_S})^{1-\beta} (1+g_t^s)^{1-\beta}}{1+g_{t+1}^N}$$

Inserting (24) yields

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = n_{t-1}(1+g_{t+1}^h)^\beta (1+g_{t+1}^{A_S})^{1-\beta} (1+g_t^s)^{1-\beta} \frac{1+n_{t-1}}{n_{t-1}(1+n_t)}$$

Combining  $n_{t-1} = n_t = n_{t+1} = \bar{n}_M$  with  $\frac{s_{t+1}}{s_t} = \left(\frac{1+g_{t+1}^{A_U}}{n_t}\right)^{1-\alpha}$  and  $1 + g_{t+1}^{A_U} = \phi_U$  gives  $1 + g_t^s = \frac{s_t}{s_{t-1}} = \phi_S^{\frac{1-\beta}{\beta}}$  in the Malthusian steady state. Additionally, in the Malthusian steady state  $1 + g_{t+1}^h = 1$ ,  $n_{t-1} = n_t = \bar{n}_M$ ,  $1 + g_{t+1}^{A_S} = \phi_S$ , therefore

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = \phi_S^{1-\beta} \phi_S^{\frac{(1-\beta)^2}{\beta}} = \phi_S^{\frac{1-\beta}{\beta}}$$

Combining the results gives the growth rate of output per capita in the Malthusian steady state as

$$1 + g_{t+1}^y = \frac{(1 + g_{t+1}^{Y_U})(1 - \theta_t) + (1 + g_{t+1}^{Y_S})\theta_t}{1 + g_{t+1}^N} = \frac{1 + g_{t+1}^{Y_U}}{1 + g_{t+1}^N}(1 - \theta_t) + \frac{1 + g_{t+1}^{Y_S}}{1 + g_{t+1}^N}\theta_t$$
$$1 + g_{t+1}^{y_M} = \phi_S^{\frac{1 - \beta}{\beta}}$$

Finally, notice that the limit case that  $s_t \to w_t^L$  only applies asymptotically as  $A^U \to \infty$ and  $A^S \to \infty$ . This implies that the expressions derived so far constitute an upper bound for savings and a lower bound for fertility. Hence, the expression of growth of income per capita also constitutes an upper bound for growth during the Malthusian phase.

**Demographic Transition** The per capita growth rate of the unskilled sector is given by

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} = \frac{n_t^{\alpha}(1+g_{t+1}^{A_U})^{1-\alpha}}{1+g_{t+1}^N}$$

Inserting (24) and re-arranging yields

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} = \left(\frac{1+g_{t+1}^{A_U}}{n_t}\right)^{1-\alpha} \frac{n_t}{n_{t-1}} \frac{1+n_{t-1}}{(1+n_t)}$$

Note that  $\frac{n_t}{n_{t-1}} \frac{1+n_{t-1}}{1+n_t}$  is non-decreasing as long as  $n_t \leq n_{t-1}$ , i.e., as long as fertility is decreasing or stable over time. If the exit from the Malthusian steady state to the demographic transition occurs in period t the expression becomes

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} = n_t^{\alpha} (1+g_{t+1}^{A_U})^{1-\alpha} \frac{1+\bar{n}_M}{\bar{n}_M (1+n_t)}$$

Inserting  $\bar{n}_M = \frac{\phi_U}{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}}, 1 + g_{t+1}^{A_U} = \phi_U$ , and re-arranging yields

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} = \left(\frac{n_t}{\bar{n}_M}\right)^{\alpha} \frac{1+\bar{n}_M}{1+n_t} \phi_S^{\frac{1-\beta}{\beta}}$$

Note that  $\left(\frac{n_t}{\bar{n}_M}\right)^{\alpha} \frac{1+\bar{n}_M}{1+n_t} > 1$  for  $n_t < \bar{n}_M$ . As a result

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} > \phi_S^{\frac{1-\beta}{\beta}}$$

as long as  $n_t \leq n_{t-1}$ .

Conversely, the per capita growth rate of the skilled sector is given by

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = n_{t-1}(1+g_{t+1}^h)^\beta (1+g_{t+1}^{A_S})^{1-\beta} (1+g_t^s)^{1-\beta}$$

where the growth rate of savings is now given by substituting (19), (4), and using (13)

$$1 + g_t^s = \frac{\frac{1 - \beta}{\beta} \frac{h(e_{t+1})}{h'(e_{t+1})} w_t^L}{\frac{1 - \beta}{\beta} \frac{h(e_t)}{h'(e_t)} w_{t-1}^L} = \frac{h\left(e_{t+1}\right)}{h\left(e_t\right)} \frac{h'\left(e_t\right)}{h'\left(e_{t+1}\right)} \frac{w_t^L}{w_{t-1}^L} = \frac{1 + g_{t+1}^h}{1 + g_{t+1}^{h'}} \left(\frac{1 + g_t^{A_U}}{n_{t-1}}\right)^{1 - \alpha}$$

Combining the results gives per capita growth in the skilled sector as

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = \frac{n_{t-1}^{1-(1-\alpha)(1-\beta)} \frac{1+g_{t+1}^h}{(1+g_{t+1}^{h'})^{1-\beta}} (1+g_{t+1}^{A_S})^{1-\beta} (1+g_t^{A_U})^{(1-\alpha)(1-\beta)}}{1+g_{t+1}^N}$$

Inserting (24) and re-arranging yields

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = (1+g_{t+1}^{A_S})^{1-\beta} \frac{1+g_{t+1}^h}{(1+g_{t+1}^{h'})^{1-\beta}} \left(\frac{1+g_t^{A_U}}{n_{t-1}}\right)^{(1-\alpha)(1-\beta)} \frac{1+n_{t-1}}{1+n_t}$$

Note that  $\frac{1+n_{t-1}}{1+n_t}$  is non-decreasing as long as  $n_t \leq n_{t-1}$ , i.e., as long as fertility is decreasing or stable over time. If the exit from the Malthusian steady state to the demographic transition occurs in period t the expression becomes

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = (1+g_{t+1}^{A_S})^{1-\beta} \frac{1+g_{t+1}^h}{(1+g_{t+1}^{h'})^{1-\beta}} \left(\frac{1+g_t^{A_U}}{\bar{n}_M}\right)^{(1-\alpha)(1-\beta)} \frac{1+\bar{n}_M}{1+n_t}$$

Inserting  $\bar{n}_M = \frac{\phi_U}{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}}, 1 + g_t^{A_U} = \phi_U, 1 + g_{t+1}^{A_S} = \phi_S$ , and re-arranging yields

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = \frac{1+g_{t+1}^h}{(1+g_{t+1}^{h'})^{1-\beta}} \frac{1+\bar{n}_M}{1+n_t} \phi_S^{\frac{1-\beta}{\beta}}$$

After the onset of the demographic transition  $g_{t+1}^h > 0$ . Additionally,  $\frac{1+\bar{n}_M}{1+n_t} > 1$  for  $n_t < \bar{n}_M$ . Lastly, note that  $1 + g_{t+1}^{h'}$  is non-increasing as  $e_{t+1}$  increases for any non-convex

human capital production function. As a result

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} > \phi_S^{\frac{1-\beta}{\beta}}$$

as long as  $n_t \leq n_{t-1}$  and  $e_{t+1} \geq e_t$ .

**Balanced Growth Path** Given  $e_{t+1} > 0$ , re-arranging (21) yields

$$\frac{1-\beta}{\beta}\frac{h(e_{t+1})}{h'(e_{t+1})} = (1-e_{t+1}) - \frac{\gamma}{w_{t+1}^H}\frac{1}{h'(e_{t+1})}$$

Inserting the result into (20) gives optimal fertility as

$$n_t = \frac{\gamma}{w_{t+1}^H} \frac{w_t^L}{h'\left(e_{t+1}\right)}$$

A balanced growth path with stable population is given by  $\frac{n_{t+1}}{n_t} = 1$ , that is

$$\frac{n_{t+1}}{n_t} = 1 = \frac{\gamma}{w_{t+2}^H} \frac{w_{t+1}^L}{h'\left(e_{t+2}\right)} \frac{w_{t+1}^H}{\gamma} \frac{h'\left(e_{t+1}\right)}{w_t^L} = \frac{w_{t+1}^H}{w_{t+2}^H} \frac{w_{t+1}^L}{w_t^L} \frac{h'\left(e_{t+1}\right)}{h'\left(e_{t+2}\right)}$$

Inserting factor prices and re-arranging yields

$$\frac{n_{t+1}}{n_t} = 1 = \left(\frac{1}{1+g_{t+2}^{A_S}}\right)^{1-\beta} \left(\frac{h'\left(e_{t+2}\right)}{h'\left(e_{t+1}\right)}\right)^{1-\beta} \left(\frac{n_t}{1+g_{t+1}^{A_U}}\right)^{(1-\alpha)(1-\beta)} \left(\frac{1+g_{t+1}^{A_U}}{n_t}\right)^{(1-\alpha)} \frac{h'\left(e_{t+1}\right)}{h'\left(e_{t+2}\right)}$$
$$= \left(\frac{1}{1+g_{t+2}^{A_S}}\right)^{1-\beta} \left(\frac{1+g_{t+1}^{A_U}}{n_t}\right)^{\beta(1-\alpha)} (1+g_{t+2}^{h'})^{\beta}$$

Inserting  $1 + g_{t+1}^{A_U} = \phi_U$ ,  $1 + g_{t+2}^{A_S} = \phi_S$ , evaluating the condition at  $\{e_{t+1}; e_{t+2}\} \rightarrow e_{max}$ , and solving for  $n_t$  yields

$$\bar{n}_{BGP} \rightarrow \frac{\phi_U}{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}}$$

The per capita growth rate of the unskilled sector is given by

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} = \left(\frac{1+g_{t+1}^{A_U}}{n_t}\right)^{1-\alpha} \frac{n_t}{n_{t-1}} \frac{1+n_{t-1}}{(1+n_t)}$$

Evaluating the expression at  $n_{t-1} = n_t = \bar{n}_{BGP} \rightarrow \frac{\phi_U}{\phi_S^{\frac{1-\beta}{\beta(1-\alpha)}}}, 1 + g_{t+1}^{A_U} = \phi_U$ , and simplifying yields

$$\frac{1+g_{t+1}^{Y_U}}{1+g_{t+1}^N} \to \phi_S^{\frac{1-\beta}{\beta}}$$

Conversely, the per capita growth rate of the skilled sector is given by

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} = (1+g_{t+1}^{A_S})^{1-\beta} \frac{1+g_{t+1}^h}{(1+g_{t+1}^{h'})^{1-\beta}} \left(\frac{1+g_t^{A_U}}{n_{t-1}}\right)^{(1-\alpha)(1-\beta)} \frac{1+n_{t-1}}{1+n_t}$$

Evaluating the expression at  $n_{t-1} = n_t = \bar{n}_{BGP} \rightarrow \frac{\phi_U}{\frac{1-\beta}{\phi_S^{\beta(1-\alpha)}}}, \ 1 + g_t^{A_U} = \phi_U, \ 1 + g_{t+1}^{A_S} = \phi_S,$  $1 + g_{t+1}^h \rightarrow 1, \ 1 + g_{t+1}^{h'} \rightarrow 1, \ \text{and simplifying yields}$ 

$$\frac{1+g_{t+1}^{Y_S}}{1+g_{t+1}^N} \to \phi_S^{\frac{1-\beta}{\beta}}$$

Combining the results gives the growth rate of output per capita on the balanced growth path as

$$1 + g_{t+1}^y = \frac{(1 + g_{t+1}^{Y_U})(1 - \theta_t) + (1 + g_{t+1}^{Y_S})\theta_t}{1 + g_{t+1}^N} = \frac{1 + g_{t+1}^{Y_U}}{1 + g_{t+1}^N}(1 - \theta_t) + \frac{1 + g_{t+1}^{Y_S}}{1 + g_{t+1}^N}\theta_t$$

$$1 + g_{t+1}^{y_{BGP}} \to \phi_S^{\frac{1-\beta}{\beta}}$$

# ONLINE APPENDIX: The Empirics of Economic Growth Over Time and Across Nations: A Unified Growth Perspective \*

Matteo Cervellati

Gerrit Meyerheim

University of Bologna CEPR University of Munich

Uwe Sunde University Munich CEPR, CESifo, IZA

September 19, 2022

<sup>\*</sup>Contacts: Matteo Cervellati, University of Bologna, Piazza Scaravilli 2, 40126 Bologna, Italy, Email: m.cervellati@unibo.it; Gerrit Meyerheim, University of Munich (LMU), Geschwister-Scholl-Platz 1, 80539 Munich, Germany, Email: gerrit.meyerheim@econ.lmu.de; Uwe Sunde (corresponding author), University of Munich (LMU), Geschwister-Scholl-Platz 1, 80539 Munich, Germany, Email: uwe.sunde@lmu.de, Tel. +49 89 2180 1280.

## A Online Appendix

	Data, 20-year panel		Baseline Model		Extension	
	(1) $(2)$		(3)	(4)	(5)	(6)
Specification:						
Full Specification	0.986	0.986	1	1	1	1
GDP p.c. $(t-1)$	0.879	0.879	0.973	0.973	0.958	0.958
k	0.873	0.873	0.997	0.999	0.993	0.995
h	0.636	0.637	0.944	0.944	0.949	0.949
Year and Country FE	0.974	0.974	0.99	0.99	0.986	0.986
Observations	269	269	342	342	342	342

Table A.1: Goodness of Fit  $(R^2)$  of Reduced Form Estimates: Data vs. Simulated Data

Variance explained  $(R^2)$  by different specifications of the empirical model based on OLS regressions. Columns (1) and (2): Full specification and variables as in Columns (1) and (2) of Table 2. Other specifications only include the respective variables indicated in the respective row as the only explanatory variables. Columns (3)-(6): Full specification and variables as in Columns (3)-(6) of Table 2.

	Western Countries:						
Dependent Variable:	Log	[Adoption]	Lag]	Log [Intensity]			
	(1)	(2)	(3)	(4)	(5)	(6)	
Invention Year-1820	$-0.0076^{***}$ (0.00)	$-0.0074^{***}$ (0.00)	$-0.0073^{***}$ (0.00)	0.000084 (0.00)	0.000076 (0.00)	0.000011 (0.00)	
Year of Demogr. Tr.		$0.0054^{***}$ (0.00)			-0.00027 (0.00)		
Follower Country			$0.36^{***}$ (0.06)			$-0.11^{***}$ (0.03)	
Trailer Country			0 (.)			0 (.)	
Latecomer Country			0 (.)			0 (.)	
Observations $R^2$	269 0.26	269 0.28	269 0.28	284 0.00	284 0.00	284 0.02	

Table A.2: Technology Diffusion Western/Non-Western Countries

	Non-Western Countries:								
Dependent Variable:	Log	[Adoption	Lag]	Log [Intensity]					
	(1)	(2)	(3)	(4)	(5)	(6)			
Invention Year-1820	$-0.011^{***}$ (0.00)	$-0.011^{***}$ (0.00)	$-0.011^{***}$ (0.00)	$-0.0053^{***}$ (0.00)	$-0.0062^{***}$ (0.00)	$-0.0059^{***}$ (0.00)			
Year of Demogr. Tr.		$0.0034^{***}$ (0.00)			$-0.012^{***}$ (0.00)				
Follower Country			$0.24^{**}$ (0.11)			0.16 (0.29)			
Trailer Country			-0.023 (0.10)			-0.35 (0.23)			
Latecomer Country			$0.33^{***}$ (0.10)			$-0.75^{***}$ (0.28)			
Observations	740	740	740	755	755	755			
$R^2$	0.43	0.43	0.46	0.12	0.23	0.27			

Panel A Columns (1) and (4) replicate the results in Table 3, columns (2) and (5) of Comin and Mestrieri (2018). Panel B Columns (1) and (4) replicate the results in Table 3, columns (3) and (6) of Comin and Mestrieri (2018). Country labels are as in Reher (2004): Forerunner countries (reference category) experienced the fertility transition before 1935, Followers between 1935 and 1964, Trailers between 1965 and 1979, and Latecomers after 1980. Robust standard errors are in parentheses. Each observation is re-weighted so that each technology carries equal weight. \*, \*\*, \*\*\* denote significance at 10%, 5%, and 1% levels.

		Dependent variable:						
	Log	[Adoption ]	Lag]	Log [Intensity]				
	(1)	(2)	(3)	(4)	(5)	(6)		
Invention Year-1820	-0.0014***	-0.0014***	-0.0014***	-0.015***	-0.015***	-0.015***		
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)		
Year of Demogr. Tr.		0.0092***			-0.013***			
		(0.00)			(0.00)			
Follower Country			0.56***			-0.80***		
			(0.04)			(0.06)		
Latecomer Country			0.75***			-1.07***		
			(0.04)			(0.06)		
Observations	1243	1243	1243	1243	1243	1243		
$R^2$	0.09	0.94	0.88	0.70	0.83	0.82		

### Table A.3: Technology Diffusion: Simulated Data

Replication of technology diffusion as in Table ?? using simulated data. The specification follows Comin and Mestrieri (2018). Adoption lag is computed as the time lag associated with reaching the level of  $A^S$  that the forerunner country reached in a given period t = 1780, 1800, ..., 1980, respectively. Intensity of use is computed as  $A^S$  in a given country relative to the corresponding levels achieved by the forerunner country at the time of invention. Country labels are as in Reher (2004): Forerunner countries (reference category) experienced the fertility transition before 1935, Followers between 1935 and 1964, Trailers (empty) between 1965 and 1979, and Latecomers after 1980. Robust standard errors clustered at the country level are in parentheses. Due to the twenty-year frequency, applying the same terminology of transitions implies that there are no trailer countries in the simulated data. \*, \*\*, \*\*\* denote significance at 10%, 5%, and 1% levels.



Figure A.1: The Dynamics of Capital Shares and Real Returns (Normalized) (a) Data: England, Crafts (2021), Table 10. Model:  $(R \cdot K)/(Y^U + Y^S)$ . (b) Schmelzing (2020), Figure IV (global rate, GDP-weighted), normalized (1700=1). Model: R and R/Y, normalized (1700=1).



(c) Capital Returns (Clark, normalized)

(d) Capital Returns (Jordà et al., normalized)

#### Figure A.2: The Real Return: Alternative Data Sources

(a) Data: England, Clark (2010), Table 7. Model: Simulated data for  $r = (R^{1/20} - 1) \cdot 100$ . (b) Data: World: Jordà et al., (2019), Figure XII (global real safe return, GDP-weighted). Model: Simulated data for  $r = (R^{1/20} - 1) \cdot 100$ . (c) England, Clark (2010), normalized (1700=1). Model: Simulated data for r, normalized (1700=1). (d) World: Jordà et al., (2019), Figure XII (global real safe return, GDP-weighted), normalized (1900=1). Model: Simulated data for r, normalized (1700=1). (d) World: Jordà et al., (2019), Figure XII (global real safe return, GDP-weighted), normalized (1900=1). Model: Simulated data for r, normalized (1700=1).



Figure A.3: Growth Relative to Frontier: Data vs. Simulation

Note: Binscatter plots, lines correspond to a quadratic fit (both panels). Annualized growth rates over twenty-year periods. (a) Data: Penn World Tables (Feenstra et al., 2015, version 9.1, years 1950, 1970, 1990, 2010.). (b) Model: extension, artificial cross-country panel, evaluated in 1960, 1980, 2000, 2020.